KWAME NKRUMAH UNIVERSITY OF SCIENCE AND TECHNOLOGY

COLLEGE OF HUMANITIES AND SOCIAL SCIENCES

SCHOOL OF BUSINESS

DEPARTMENT OF ACCOUNTING AND FINANCE

NINO SI

IMPACT OF DIVIDEND POLICY ON THE PERFORMANCE AND VALUE OF

FIRMS LISTED ON THE GHANA STOCK EXCHANGE

BY

ABIDATU AWUDU ABDULAI

(PG20730922)

A THESIS SUBMITTED TO THE DEPARTMENT OF ACCOUNTING AND

FINANCE, KWAME NKRUMAH UNIVERSITY OF SCIENCE AND

TECHNOLOGY IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE

OF

MASTER OF BUSINESS ADMINISTRATION IN FINANCE

SEPTEMBER, 2023.



DECLARATION

I hereby declare that this submission is my work towards the Master of Business Administration
in Finance and that, to the best of my knowledge, it contains no material previously published
by another person nor material which has been accepted for the award of any other degree of
the University, except where due acknowledgement has been made in the
text.

Abidatu Awudu Abdulai		
(PG20730922)	Signature	Date
		1
Certified by :	ELCR	113
Prof. Michael Adusei		
(Supervisor)	Signature	Date
NH HAND		BADH THIS
Prof. Kingsley Owusu Appiah	SANE NO	
(Head of Department)	Signature	Date

DEDICATION

I dedicate this work to my husband and children for the support and love.



ACKNOWLEDGEMENT

I am most grateful to the Almighty Allah for his grace and mercy. I am thankful for the guidance and impeccable supervision I received from Dr. Michael Adusei as my supervisor. His time and expertise have been a tremendous help towards the successful completion of this research work and have given a better shape to the output of the work. Not forgetting my parents Mr and Mrs AWUDU Jambo, my husband ISSAH Abubakar, and my siblings especially Mabruk Mohammed, I am most grateful.



ABSTRACT

This study examines the impact of dividend policy on the financial performance and value of listed firms in Ghana. The study employs purposive sampling technique to select 15 firms listed on the Ghana Stock Exchange and uses secondary data from the audited financial statement of the firms that meet the inclusion criteria of the study. The study uses an ordinary least square regression model to establish the relationship between dividend policy and financial performance and firm value and simultaneously examines the effect of other control variables (leverage, liquidity, and firm size) on financial performance and firm value. From the findings of the study, the researcher concludes that dividend policy has a significant impact of the financial performance of listed firms in Ghana, thus firms perform well when dividend shareholders receive increases. More so, the findings also indicate that larger firms tend to perform better than firms of small sizes. Also, the researcher concludes that firms with higher levels of liquidity tend to perform better than firms with lower levels of liquidity. Furthermore, firms with a higher degree of leverage, thus using debt to finance their operations, perform better financially. Notwithstanding this, on the basis of the findings of this study, the researcher concludes that firms with higher degree of leverage experience a significant decrease in the valuation of the firm. However, liquidity does not have any significant effect on the value of listed firms in Ghana. Financial managers need to carefully evaluate the level of leverage that is right for their business, weighing the possible advantages of debt financing operations against the impact on the firm's valuation. For the firm to avoid taking on excessive risk and to maintain its overall financial stability, a careful approach to controlling leverage is essential. Financial managers should evaluate the company's financial standing and, if feasible, explore raising dividend payments. Since dividend policy has a substantial impact on the financial performance

of listed corporations in Ghana, this may help the company attract investors and possibly improve its financial performance.

TABLE OF CONTENT

DECLARATION					i
DEDICATION	K.			<u> </u>	
ii					
ACKNOWLEDGEMENT	••••••			•••••	iii
ABSTRACT					
iv	5				
TABLE OF CONTENT					V
LIST OF TABLES					ix
LIST <mark>OF FIGURES</mark>					x
	à.	10	D,	£Z	7
CHAPTER ONE					1
1.0 INTRODUCTION					1
1.1 Background Information.					1
1.2 Problem Statement					
1.3 Aim and Objectives of the	e Study				
1.3.1 Aim of Study				<u>y</u>	4
1.3.2 Specific Objectives					4
1.4 Research Questions					4

1.5 Significance of Study	4
1.6 Scope of Study	5
1.7 Brief Methodology	5
1.8 Organization of Chapters	6
CHAPTER TWO	7
LITERATURE REVIEW	7
2.0 Introduction	7
2.1 Conceptual Review	7
2.1.1 Dividend Policy	7
2.1.2 Firm Performance	8
2.1.3 Firm Value	9
2.2 Theoretical Review	9
2.2.1 Dividend Irrelevance Theory	10
2.3 Empirical Review	13
2.4 Conceptual Framework	24
CHAPTER THREE	
METHODOLOGY	
3.0 Introduction	25
3.1 Study Design	25
3.2 Study Population and Sample Size	26

3.2.1 Population	26
3.2.2 Sampling Technique and Sample Size	26
3.3 Data Collection Technique	26
3.4 Study Variables	27
3.4.1 Dependent Variables	27
3.4.2 Independent Variables	27
3.4.3 Control Variables	27
3.5 Estimation Technique	
3.6 Model Diagnostics	29
3.6.1 Multicollinearity Test	30
3.6.2 Normality Test	
3.7 Data Analysis and Presentation	
CHAPTER FOUR	
RESULTS AND DISCUSSION	
4.0 Introduction	
4.1 Results	
4.1.1 Descriptive Statistics	
4.1.2 Test for Normality	
4.1.3 Multicollinearity	
4.1.4 Model Summary and Test for Autocorrelation	

4.1.5 Analysis of Variance	37
4.1.6 Regression Analysis of Study Variables	
4.2 Discussion	43
CHAPTER FIVE	49
SUMMARY OF FINDINGS, CONCLUSION AND RECOMMENDATIONS	49
5.0 Introduction	49
5.1 Summary of Findings	49
5.2 Conclusion	50
5.3 Recommendations	50
REFERENCES	52
APPENDICES	



LIST OF TABLES

Table 3.1: Definition and source the operational variables of the study
Table 4.1: Descriptive statistics of study variables 33
Table 4.2: Kolmogorov-Smirov test of normality 34
Table 4.3: Value inflation factor (VIF) and tolerance values for the explanatory variables 35
Table 4.4: Correlation matrix of study variables 35
Table 4.5: Durbin-Watson test for autocorrelation 36
Table 4.6: Analysis of Variance for the model with ROA as the dependent variable
Table 4.7: Analysis of Variance for the model with ROE as the dependent variable
Table 4.8: Analysis of Variance for the model with TQ as the dependent variable
Table 4.9: Relationship between dividend policy, liquidity, leverage and size, and the
performance and value of listed firms
WJ SANE NO

LIST OF FIGURES

Figure 2.1: Conceptual framework for the study	
--	--



CHAPTER ONE

1.0 INTRODUCTION

1.1 Background Information

A firm's payout of profits to its shareholders is known as a dividend. A firm is able to distribute a portion of its profit as a dividend to shareholders when it generates a profit or surplus (Sondakh, 2019). Retained earnings, or the profit that remains after the dividend, will be reinvested in the future. Many have long disputed the dividend and how it impacts how investors assess stocks (Amidu, 2007a; Shah & Noreen, 2016). For a company's stock price to remain stable and rise, as well as to boost investor returns, its financial performance and value are crucial.

A finance manager's top priority is to maximize shareholder wealth, which serves as return on investment and is reflected in the firm's value. Dividend decisions are crucial to a company's valuation and performance, which literally translate to capital gains in share prices (Selvam *et al.*, 2016). Despite the inverse relationship between dividend and earnings ratio, return is made up of two parts: dividends and rising stock prices (capital gains) (Adam *et al.*, 2017; OforiSasu *et al.*, 2017). Dividends and retained earnings both aim to maximize shareholders' wealth; retained earnings are used to finance viable expansion projects, while dividends give stakeholders more negotiating power (Eka, 2018a).

Increasing a company's profitability and cash flow is associated with maximizing shareholder wealth (Obaidat, 2019). The idea behind dividend policy is to maximize shareholder wealth by paying out long-term dividends or raising share prices for capital gains (Husain & Sunardi, 2020; A. Rahman, 2018). Even though it's one of the most difficult things to understand, dividend policy is nonetheless crucial to a company's success. The conflict over how much

must be given in dividends to shareholders and how much must be kept for investment and growth is very important since it affects how investors act generally (Nkn, 2018).

Numerous academics from all around the world have conducted studies on how the dividend policy affects a company's financial performance and value. According to several academics, this subject presents one of the most difficult study problems (Eka, 2018b; Onanjiri & Korankye, 2014). Others believe that the firm's plan for distributing revenue to shareholders, known as its dividend policy, is more than just a financial transaction (M. N. Khan *et al.*, 2016; Obayagbona & Ogbeide, 2018).

Despite different research, the earlier studies have shown that there are disparities in the way that dividend policies affect a company's financial performance and value. According to some academics, dividend policy has a considerable and favorable impact on financial performance and firm value (Pradana, 2021; Rizqia & Sumiati, 2013). Others claimed that the dividend policy had a big, negative impact on the performance of the company (Onanjiri & Korankye, 2014). Inconsistencies exist not just between study years but also between countries (Kim & Kim, 2020), and even between different economic sectors within a single nation (Nguyen Trong & Nguyen, 2020; Shao *et al.*, 2010). Using path analysis for the variable profitability of company value with dividend policy as an mediating variable, it was discovered that profitability has a significant positive influence on dividend policy and dividend policy has a huge positive effect on firm value (Husain & Sunardi, 2020). However, Sugiastuti *et al.*, (2018)'s conclusion demonstrated that business value is not significantly increased by profitability or dividend policy.

Despite the significance of dividend policy, little is still known about how it affects the performance and value of a company in Ghana. As mentioned above, numerous research have investigated the impact of dividend policy on business performance and value, but the findings

are still contradictory and inconclusive. This study aims to examine the effect of dividend policy on the performance and value of enterprises utilizing firms listed on the Ghana Stock

Exchange, taking into account the gaps in the available literature.

1.2 Problem Statement

While investors are curious in the dividend value, researchers and businesses are constantly worried about dividend payments. There have been some questions raised over the amounts of dividend from income that should be sent to shareholders, including whether they should receive a cash dividend, a stock dividend, or nothing at all (Jabbouri, 2016; Ozuomba *et al.*, 2016). The effect of dividend distribution on firm value and performance has been investigated extensively (Husna & Satria, 2019; Ngo & Dang, 2016; Seth & Mahenthiran, 2022), yet the findings are still contradictory and ambiguous.

It is impossible to overstate the competing interests of shareholders on dividend policy; every rational shareholder will consistently demand that a bigger dividend be paid, regardless of the firm's investment choices. As seen by the rise in stakeholder value, finance managers are faced with a difficult issue regarding how to balance the two decisions (dividend and investment) (Banerjee, 2015; Yuniningsih *et al.*, 2019). Most of the studies that assess the connection between dividend policy and firm performance and value have used data from developed economies. Evidence from developing economies such as Ghana is yet to clearly elucidate the effect of dividend policy on financial performance and value of firms. This leaves a gap to be bridged. This study responds to this using the Ghana Stock Exchange's data. The choice is informed by data availability.

1.3 Aim and Objectives of the Study

1.3.1 Aim of Study

The study seeks to examine the impact of dividend policy on the performance and value of firms

listed on the Ghana Stock Exchange

1.3.2 Specific Objectives

The specific objectives of the study are:

1. To examine the impact of dividend policy (dividend per share) on the performance (ROE

and ROA) of firms listed on the Ghana Stock Exchange

- To assess the impact of dividend policy on the value of firms listed on the Ghana Stock Exchange
- 3. To examine the impact of other variables (firm size, liquidity and leverage) on the performance and value of firms listed on the Ghana Stock Exchange.

1.4 Research Questions

- 1. Is there an association between dividend policy and the performance of firms listed on the Ghana Stock Exchange?
- 2. Is there an association between dividend policy and the value of firms listed on the Ghana Stock Exchange?
- Do other factors such as firm size, liquidity, and leverage influence the performance and value of firms listed on the Ghana Stock Exchange?

RAD

1.5 Significance of Study

Empirically, the study will supplement the current literature on the subject matter as it would enable researchers and academicians in the field to better appreciate the effect of dividend policy on the performance and value of listed firms. It will also provide information on the mediating effect of liquidity, financial leverage, and firms size on the performance and value of firms considering the existing dividend policy of the listed firms The findings of the study will be useful to the management of the organizations understudied and those of firms in other industries where decision making on dividends is critical.

1.6 Scope of Study

The study will focus on dividend policy and its impact on the performance and value of firms listed on the Ghana Stock Exchange. Purposeful sampling as the mode of sampling for the study. Purposeful sampling strategy will be used because the study will involve only firms listed on the Ghana Stock Exchange and have remained listed for the last 10 years. The population of the study will comprise all listed firms. The study will make use of pre-designed data collection sheets to aid the researcher in collecting the need data from the financial records of the listed firms.

1.7 Brief Methodology

In terms of methodology, the study will be limited to the use of the quantitative methodology. A firm is chosen provided the firm remains listed in the market for at least two-thirds of the period and paid dividends for at least ten years, firm-year observations from financial reports of the firm on the website of the Ghana Stock Exchange is examined. This is to ensure a majority of paying dividend-paying firms are included in the study with the assumption that dividends are relevant to corporate financial managers. Panel data for the period 2010-2021 will be used.

1.8 Organization of Chapters

The study is organized as follows: Chapter One covers the introduction of the study, the background to the study, statement of the problem, aim, objectives of the study, research questions, and significance and brief methodology of the study. Chapter Two covers literature review. Chapter three covers the methodology and estimation procedures adopted for the study. Chapter four covers the data analysis and Chapter Five presents a summary of findings, conclusion and recommendations.



CHAPTER TWO

LITERATURE REVIEW

2.0 Introduction

This chapter reviews literature important to the subject of this study. It covers conceptual review, theoretical review and empirical review. These three reviews are used to develop the conceptual framework of the study.

2.1 Conceptual Review

2.1.1 Dividend Policy

The distribution of earnings between retained earnings and payouts to shareholders is based on the dividend policy. The importance of dividend strategy is dependent on how well the company's management can balance growth with dividend payments to shareholders. As a result, dividend policy plays a significant part in the firm's value. Companies could use fixed or residual dividend policies (Oppong, 2016).

Dividends are the incentives given to shareholders of corporations in the form of money, stock, or other items (Aryati, 2017). The board of directors of a corporation determines the dividend, which must be approved by the shareholders. Dividend payments are not required, but they are a well-liked practice for rewarding shareholders with a portion of the company's leftover profits. The money that is still available after fulfilling other commitments, such as paying creditors, is referred to as residual rewards (Olang & Grace, 2017). According to (Bataineh, 2021), the company must take into account two major goals when establishing a dividend policy: first, to have enough money to pay dividends; and second, to maximize the wealth of the shareholders.

2.1.2 Firm Performance

Financial performance is a gauge of how effectively a company can use resources from its main line of business to increase profits and expand its operations (Jihadi *et al.*, 2021). There are numerous techniques to measure financial performance, but they should all be combined (Akisik & Gal, 2019). Financial ratios from financial statements are considered a trustworthy source of data to evaluate financial achievement Restianti & Agustina, (2018). Net earnings, which are divided into retained earnings and dividends, are another metric used to assess financial performance. The company's retained earnings can be invested in and used as a source of long-term capital (Triani & Tarmidi, 2019). The shareholders receive dividend payments to increase their wealth because they invested their money to do so.

A performance metric known as profitability focuses on the relationship between revenues and costs as well as the level of profits concerning the amount of capital invested in the company (M. M. Rahman *et al.*, 2015). The operating profit margin, net firm income, return on equity (ROE), and return on total assets (ROA) are the four most frequently cited indicators of a company's profitability (Enekwe *et al.*, 2015). With each cedi of shareholder stock, a company's return on equity (ROE), a measure of profitability, determines how much profit it makes. ROE is calculated using the formula ROE = Net Profit/Shareholders' Equity. Return on equity is also known as return on net worth. Return on assets is a measure of a company's profitability in relation to its total assets (Sausan *et al.*, 2020; Utami, 2017). The ROA reveals how well management uses its resources to produce profits. ROA is expressed as a percentage and is calculated by dividing a company's annual earnings by its total assets. ROA is calculated as Net Profit / Total Assets.

2.1.3 Firm Value

The total value of all a company's financial rights can also be thought of as its value. The worth of the firm is based on the ongoing anticipation of all the predictable future cash flows that the assets will generate, discounted at the corporation's weighted average cost of wealth (Chowdhury & Chowdhury, 2010). According to Pandey (2005), a company's worth is equal to the sum of its financial instruments' value. The total cash flow that assets generate should equal the money streams obtained by the necessary claims. The issue of exploitation gets considerably more complicated in a variety of circumstances where the company's earnings differ. The dividend streams that the stockholder will get throughout the company's existence, discounted back to the current value, can be used to determine the value of the company (Parkinson & Waweru, 2010).

A key factor in an investment criterion is a firm's value. Different metrics, including net sales, paid-up capital, total assets, capital employed, and others, can be used to determine a company's value (Sharma, 2011). It is assumed that a company's worth will account for both its tangible and intangible assets. Tobin's Q is a widely used tool for determining a company's value. Tobin Q often represents a ratio of a company's market worth to the cost of replacing its assets (Taslim, 2013). Tobin Q compares market-based measurements to book value when determining a company's value. A corporation is said to be valuable according to its q proposition if investment returns outweigh investment costs (Taslim, 2013).

2.2 Theoretical Review

20

Several theories have been proposed regarding the dividend policy. Several scholars have examined these theories over the years. To examine and give a theoretical explanation of the impact of dividend policy on a firm's valuation and performance, the current research explores the dividend irrelevancy theory postulated by Miller & Modigliani, (1961).

BADW

2.2.1 Dividend Irrelevance Theory

One of the most important dividend theories was provided by Modigliani & Miller in 1961, and although being created more than 50 years ago, it is still regarded as one of the most reputable theories. The theory's presentation in the essay "Dividend policy, growth, and the valuation of shares" set a new standard and altered the perception of dividends among academics and practitioners alike. The prevalent belief prior to the publication of Modiglianidividend Miller's irrelevance argument was that dividends were closely tied to stock valuation (Tanushev, 2016). As the name of the theory suggests, under ideal capital markets, the dividend policy is independent of the firm's price and does not depend on whether the company pays out high or low dividends.

The theory makes the following assumptions: there are no taxes, or the tax rates on cash dividends and capital gains are equal; there are no transaction costs for the process of selling or buying shares, so if the investor needs cash, he or she will be able to sell his or her shares without losing commissions and fees instead of cash dividends; the investor is rational in his or her decisions; and there are no agency costs, meaning that companies will not have to pay commissions. The theory also assumes that there is no information gap and that the company operates in a full and efficient market. A full and efficient market is one in which information is simultaneously available to all participants at no cost, reflects this information, and is influenced by it at the time it is provided. Last but not least, the theory makes the supposition that all investors' expectations and knowledge of the company's performance in the future are uniform.

According to MM, given a world in which investor behaviour is not irrational, that is, where investors constantly desire to have more wealth rather than less, regardless of whether it comes from a cash or capital gains, where there is "perfect certainty" on investors' behalf that they will invest and their returns are certain, and where the market is perfect, that is, where no single

entity can influence the market, the firm value does not depend upon the dividend policy, The equation below, which was provided by Miller & Modigliani, (1961), proves this assertion:

$$V(t) = n(t)p(t) = \frac{1}{\rho(t)} [X(t) - I(t) + V(t - 1)]$$

The basic assumption of their argument was that the value of a company is decided by its earning potential and is based on the profits it makes by making the finest investments feasible (investment policy). When the investment decision is made, the payout policy loses significance to the firm's value. This is done to ensure that the residual (the difference between investments and earnings) accurately represents the net payout. In other words, the company's worth is decided by the income generated by its assets, not by how this money is distributed between retained earnings and dividends. Dividend policy is meaningless from the standpoint of the investor because prudent equity purchases and sales can mimic the desired stream of payments. With a balancing adjustment to the number of shares outstanding, a firm can change its dividends to any level. The authors contend that dividend policy is irrelevant as it cannot alter the shareholder's wealth, to the shareholder hence, investors will follow whatever dividend policy there is and maybe disregard the premium.

The dividend irrelevance theory has been supported by several authors. Hansda *et al.*, (2020) nvestigate the relationship between dividend policy and firm value in the context of the financial crisis. The analysis is based on data from 500 BSE-listed firms from 2001 to 2017. It is used the dynamic panel regression using a two-step approach Generalised Method of Moments (GMM). The findings indicate that dividend policy has no effect on firm value. Another study has been conducted by Hasan *et al.*, (2015) to check an association between dividend policy and earnings per share in the textile and energy sectors of Pakistan. The findings of logarithmic regression revealed that dividend payout ratio had a negative influence

on firm earnings regardless of industry. Rizwan *et al.*, (2016) conducted a recent study on companies listed on the Pakistan stock exchange. According to their research, dividend policy has no effect on return on equity. Another study conducted by Tahir *et al.*, (2016) tested the effect of dividend policy on firm performance and concluded that there is a positive significant association between the performance of firms and their respective dividend payout policy.

The irrelevance argument of Miller and Modigliani (MM) has been questioned by several academics who have studied the impact of dividend policy on firm value and performance (Ayunku & Apiri, 2020; Budagaga, 2017 Singh & Tandon, 2019). They contended that the assumptions made by MM do not hold in an imperfect market. As an illustration, while investors pay transaction costs every time they sell or buy shares, firms do so while trying to obtain new money. Companies in Ghana must cover brokerage costs or underwriting expenses when issuing new shares. In Ghana, capital gains are tax-free, however dividends are subject to a 10% withholding tax. Furthermore, because insiders have access to more knowledge than outsiders, the market does not accurately reflect all facts. These obviously prove that dividend policy has a significant impact on the pricing of shares in a flawed market like the Ghana Stock Exchange.

Transaction costs, ambiguity tenacity, taxes, accounting manipulation, behavioural finance, cash flow, agency costs, and signalling are a few of the aspects that have been discovered in prior empirical investigations about the relevance of dividend policy of corporations (Abor & Amidu, 2006; Allen & Michaely, 2003; Amidu, 2007b; Ayunku & Apiri, 2020; Budagaga, 2017). Transaction cost is a logical justification for dividends. Buy dividend-paying stocks and take the dividend payments, or buy non-dividend-paying equities and periodically sell a portion of your portfolio, if you wish to get a regular income from your securities holdings. The transaction costs connected with collecting dividends may be much lower for a small individual

investor than the transaction expenses involved with selling a portion of the equities (Busse *et al.*, 2021).

2.3 Empirical Review

The impact of dividend policy on a company's performance and value has been a contentious issue with divergent opinions from academics. The return on assets (ROA) and ROE are a metric used in finance to assess how well assets have been used to produce profits. A higher Return on Assets (ROA) denotes a company's performance because of the higher rate of return on investment (Suardana *et al.*, 2018). Profits before interest expenses for a given period are what is referred to as return on equity (ROE), which is a dividend of total shareholder equity.

Khan *et al.*, (2018) use the ordinary least square method to create the connection between dividend policy and the performance of firms listed in Pakistan. The study examines the effect of dividend policy on company performance using data from 9 cement manufacturing companies for the years 2012 to 2016. Dividend per share, earnings per share, business size, and leverage serves as proxies for dividend policy, whereas the ROE serves as company performance proxy. The study discovers a significant positive association between earnings per share and ROE in Pakistan's cement industry. Dividend per share financial leverage and return on equity do not, however, have a meaningful correlation. Return on equity and firm size have a significant association with the financial performance of the selected cement manufacturing companies.

Narang, (2018) investigates the relationship between the dividend payout and financial performance among listed companies on the National Stock Exchange (NSE). The study employs correlation and regression analysis as its methodology to analyse data from 2012 to 2017. Earnings per share, dividend payout ratio, and price-earnings ratio are the metrics used to gauge dividend policy, while return on equity and return on the asset are used to gauge

business performance. The study discovers that return on equity and return on an asset does not correspond with earnings per share, dividend payout ratio, or price-earnings ratio.

In a related study by Ugwu *et al.*, (2020), the researchers examine dividend policy's impact on firm financial performance among consumer products companies in Nigeria from 2015 to 2019. The study employs OLS and correlation matrix methods and chooses 10 consumer food companies that are listed on the National Stock Exchange. Dividend policy is proxied by dividend payout ratio (DPR), earnings per share (EPS), and dividend per share (DPS), and ROE serves as a gauge of a firm performance. The study makes use of firm size (FSZ) and financial leverage (FLV) as control variables. According to the results, the researcher concludes that DPS and return on equity have a statistically significant association. The relationship between the variables DPR, EPS, and FSZ and return on equity is statistically insignificant. FLV does, however, have a weak and insignificant association with return on equity.

Nguyen *et al.*, (2021) investigate the effect of a company's dividend policy on its financial success. The study uses ROA, ROE, and Tobin's Q as dependent factors and dividend rate and dividend choice, as independent factors, the article analyzes the research gap. The source of data for the study is the financial records of the 450 listed companies on the Vietnam Stock Exchange between the years 2008 and 2019. The results of the study demonstrate that despite improving market expectations for businesses, paying dividends has a negative impact on Vietnamese companies as measured by accounting-based performance. The study also finds that Vietnamese companies' low dividend rates have a impact on accounting-based performance that is favourable yet has an adverse effect on market expectations.

A company's ability to provide enough cash to fulfil its short-term obligations is measured by its liquidity, which is stated as a ratio of current assets to current liabilities (debt) (Rashid, 2018). The fast ratio and the current ratio are the two most popular liquidity ratios. The current

ratio shows the extent to which assets that are anticipated to be turned into cash in a time frame roughly matching the maturity of the liabilities are used to pay short-term creditors' claims. A company's current liabilities are sums of money that are due within the next year or less. A study by Li *et al.*, (2020) on the association between liquidity and firm performance, uses panel data from the publicly available annual reports of 15 entities covering the years 2008 to 2017. The study uses a random effect generalized model to assess the association between liquidity and firm performance. The Return on Equity (ROE) of a corporation is strongly impacted negatively by liquidity, according to estimates from a random effect generalised least squares (GLS) regression, but is only marginally impacted favourably when ROE is measured using the cash flow ratio.

A related study in Kenya by Akenga, (2017) examines how the performance of companies listed on the Nairobi Securities Exchange is impacted by liquidity. The study uses current ratio, cash reserves and debt ratio as representations for liquidity and ROA as a representation for financial performance, whiles employing correlational analysis and a multiple regression model to establish the relationship between the study variables. According to the study's findings, cash reserves and the current ratio have a big impact on ROA. However, ROA, a metric of the financial success of the listed companies on the Nairobi Securities Exchange, is unaffected by the debt ratio in a major way.

Bibi & Amjad, (2017)'s study investigates the relationship between a firm's liquidity and profitability. In order to conduct an empirical analysis of the relationship between liquidity and business profitability, data from 50 companies listed on the Pakistani Karachi Stock Exchange were gathered. The research employs the use of panels based on secondary data from 2007 to 2011. Utilizing net operating income and return on assets, one can gauge a company's profitability. The cash gap in days and the current ratio are used to assess the firm's liquidity.

Regression analysis is used in the study to analyse how liquidity affects profitability. To determine the significance of particular model variables, incremental tests are run. According to the results of correlation and regression analysis, the cash gap and return on assets have a strong inverse link, but the profitability and current ratio have a substantial inverse relationship. Kong *et al.*, (2019) explore the relationship between liquidity and the financial performance of listed non-financial firms in Ghana. The study uses correlational analysis as it sought to examine the relationship between liquidity and the firms' performance. The study employs the Pearson Product-Moment Correlation Coefficient technique of data analysis and it shows that liquidity has a significant relationship with the firms' financial performance as measured by ROA, but an insignificant relationship with the firms' financial performance as measured by ROE and ROCE.

The size of a firm has a variety of effects on its financial success. In comparison to small businesses, large corporations can take advantage of economies of scale and breadth, making them more effective (Wells, 2016). Additionally, small businesses might not have as much clout as big businesses, making it challenging for them to succeed, especially in marketplaces with fierce competition. On the other side, as businesses become bigger, they may experience inefficiencies that worsen their financial performance. Therefore, the exact relationship between size and performance is unclear according to theory.

Olawale *et al.* (2017) also investigates the effect of firm size on the performance of firms in Nigeria using panel data set of 12 non-financial firms operating in Nigeria from the period 2005-2013 and analyzing the panel data using a pooled regression model, fixed effect model and random effect model to identify the relationship between firm size and the performance of firms listed on the Nigeria stock exchange. The result of the study reveals that firm size in terms of the total asset has a negative effect on performance while in terms of total sales firm size has a positive effect on performance.

In Eyigege, (2018)'s study, the author examines the influence of firm size on the financial performance of deposit money banks quoted on the Nigerian stock exchange. The study selects five deposit money banks using the Taro Yemeni sampling technique to represent the entire banking industry in Nigeria. The study uses a log of total assets as a proxy for firm size and ROA as a proxy for financial performance. Pooled OLS regression and fixed effect/random effect regression for panel regression serves as the statistical tools for analysing the panel data from the financial records of the selected banks. The finding of the study indicates that firm size insignificantly negatively influenced financial performance as a result of diseconomies of scale.

The study by Musah & Kong, (2019) examines the performance of non-financial companies listed on the Ghana Stock Exchange and the relationship between firm size and performance. The uses return on assets (ROA), return on equity (ROE) and return on capital employed (ROCE) as a proxy for financial performance. The study takes a quantitative approach to gather balanced secondary panel data from the sampled companies' audited and published annual reports for the years 2008 to 2017. The research uses correlational analysis. The research uses the Pearson Product-Moment Correlation Coefficient a method of data analysis to determine the bivariate relationships between firm size and financial performance. According to the study's correlational estimates, size is considerably correlated favourably with the firms' ROAbased financial success. However, there is a negligible negative correlation between company size and ROE and ROCE.

The debt-to-equity ratio (debt/equity ratio) is a way to quantify debt leverage and it demonstrates the extent to which a company uses borrowed funds (Atidhira & Yustina, 2017).

17

High-leveraged businesses run the risk of going bankrupt if they are unable to fulfil their loan payments, and they may also have trouble finding new lenders down the road. However, leverage is not always a bad thing; it can boost shareholders' returns on investment and make the most of the tax benefits linked to borrowing (Ahmed *et al.*, 2018). A firm's financial status or the level of leverage can be determined by using indicators like the debts to equity ratio and the debts to total assets ratio (Enekwe *et al.*, 2014; Widyastuti, 2019). When it comes to raising production levels, increasing shareholder value, or acquiring new assets, financial leverage is crucial for every firm.

Bunyaminu *et al.*, (2021) investigate the influence of financial leverage on the profitability of recapitalized banks in Ghana from 2008 to 2017. The study employs ROA as a proxy for profitability from secondary data obtained from the financial statement of the banks. Based on the estimation methodologies for random effects and fixed effects, the results of the study reveal that debt which is defined as the ratio of total debts to total assets exerts a significant negative effect on banks' profits regardless of the proxy of profitability.

In Gathara *et al.*, (2019), the study's goal is to look into the impact of leverage on the performance of companies listed on the Nairobi Securities Exchange (NSE). The study uses a causal or explanatory research design to collect and analyse quantitative secondary data for 30 selected companies from the period 2007 - 2015. The study further uses multivariate tests using a panel data m6odel to examine the effects of the independent (leverage) variable on a company's financial performance. The study reports that Leverage had a significant positive effect on the financial performance of selected companies listed at NSE, Kenya.

The research of Iqbal & Usman, (2018) Examine the relationship between financial leverage and the performance of Pakistani Textile Composite Companies. Pakistan Textile Composite Companies listed on the PSX (100-index) are chosen. From 2011 to 2015, 5-year data is collected, and the top 16 organisations are chosen as a sample. To identify the outcomes, descriptive statistics, correlation analysis, and a regression model were used. Financial leverage has a negative and substantial effect on business ROE, but a positive and significant effect on firm ROA, according to the findings. According to the study, a high interest rate and a large quantity of debt reduce the value of equity and have a negative influence on business performance. The quantity of debt, on the other hand, has a positive impact on the firm's ROA. Financial leverage has a beneficial impact on firm performance when the quantity of debts does not exceed the amount of equity, according to the findings.

The decision regarding the company's dividend policy is crucial, so how managers approach this decision-making process, as well as whether or not they keep an eye on a specific set of rules or detailed plans to implement these adoptions, will affect the firm's worth. Tobin's Q as a company value indicator has been taken into account by Amidu, (2007), Dang *et al.*, (2021) and Kadim *et al.*, (2020). This ratio assesses the value offered by financial markets to any management or organisation as a growing business. Tobin's q also demonstrates how far a company's value can be created in relation to the quantity of capital invested. The bigger the value of Tobin's q, the better the company's growth chances. (Setiyawati *et al.*, 2017). When market value and replacement cost are equal, equilibrium is achieved. At its most basic level, the Q Ratio best expresses the relationship between market valuation and intrinsic value. It is a technique for analysing if a certain company or market is overvalued or undervalued. Tobin's Q is anticipated to have a positive association with the dividend policy

Ofori-Sasu *et al.*, (2017)'s research examines the effect of dividend policies on shareholder value amongst firms listed in Ghana. The study makes use of pooled OLS for the years 2009 to 2014, panel regression is used to analyse data from Ghana Stock Exchange-listed companies.

The study uses secondary data from the financial records of the listed firms on the Ghana Stock Exchange. The study discovers a positive relationship between dividend per share and shareholders' value. However, the study reports a negative and strong association between dividend yield and shareholders' value.

Emeni, (2017) explores the influence of stock dividend policy on company market value. The researcher uses 142 businesses listed on the Nigerian Stock Exchange (NSE) for the fiscal years 2002 to 2011 in his study on the correlation between a company's market value and its cash and stock dividend policies. The technique of ordinary least squares is used to analyze the data gathered from the financial statement of the selected business. The findings demonstrate that (a) cash dividends and investment philosophy have a weak but significant negative relationship with the market value of companies listed on the NSE, while (b) stock dividends and earnings have a strong and positive relationship with the market value of companies listed on the NSE.

The study conducted by Lumapow & Tumiwa, (2017) examines the effects of business size, productivity, and dividend policy on firm value. The researchers use purposive sampling to choose the sample, which is manufacturing companies listed on the Indonesia Stock Exchange (IDX) between 2008 and 2014. Panel data regressisson with a Random Effect Model (REM) approach is the analysis method used in this study. The findings of their tests indicate that the dividend policy is negatively and significantly correlated with firm value, that firm size is positively and significantly correlated with firm value, and that firm productivity is positively and significantly correlated with firm value.

Hansda *et al.*, (2020) investigate the connection between dividend policy and corporate value. Data from 500 firms listed on the BSE from 2001 to 2017 serves as the foundation for the investigation. The data is analysed using the two-step dynamic panel regression Generalized Method of Moments (GMM). The results demonstrate that dividend policy had no significant impact on firm value. The study does note, however, that the association between dividend behaviour and business value is altered by the financial crisis.

The study by Odum *et al.*, (2019) investigates the impact of dividend payout ratio on the value of a corporation. The study also investigates other characteristics (profitability, leverage policy ratio, dividend policy ratio, cash holding, and firm size) that affect firm value while using companies listed on the Nigerian stock exchange. In order to analyse the data, the study used Panel Ordinary Least Square Regression Techniques. The analysis exclusively includes breweries and beverage companies that were listed on the Nigerian stock exchange between 2007 and 2016. The results reveal that the profitability ratio and leverage ratio have a positive and considerable impact on the firm's worth. This shows that only Firm Leverage and Profit after Tax are major drivers driving firm value in both breweries and beverage companies among Nigerian listed companies. The data also show that there is no significant association between dividend pay ratio and company valuation among the study's sample firms.

Azaro *et al.*, (2020)'s research aims to establish the effect of firm size, leverage, profitability, and price-earnings ratio on firm value. The study uses 11 manufacturing companies sector consumer goods industry listed on the Indonesian Stock Exchange from 2013-2017 as its sample size and employs purposive sampling as its sampling technique. This study is quantitative research with a descriptive approach which employs multiple linear regression as the study's analytical technique. The results of the statistical analysis show that partially variables of profitability and price-earnings ratio have a significant influence on firm value. Meanwhile, firm size and leverage do not have a significant influence on firm value.

Markonah *et al.*, (2020)'s research is to ascertain the impact of profitability (ROA), leverage (DER), and liquidity (CR) on corporate value (PBV) in food and beverage manufacturing

21

businesses listed on the Jakarta Stock Exchange. The study adopts a regression analysis method with a fixed effect model using panel data from 14 publicly traded firms from 2010 to 2016. The F test results reveal that profitability as measured by ROA, leverage (DER), and liquidity (CR) effectively account for the influence on firm value (PBV). According to the findings of the study, profitability (ROA) and leverage (DER) have a substantial effect on the company value variable (PBV), however liquidity (CR) has no significant effect on the company value variable (PBV).

In a similar study, Sondakh, (2019) investigates the impact of dividend policy, liquidity, and firm size on firm value in the financial services sector industries listed on the Indonesian Stock Exchange. The study comprises 12 firms that met the sampling standards by adopting purposive sampling from 99 financial services firms between 2015 and 2018. The panel data gathered from the financial statements of the selected organisations is analysed using multiple linear regression in this study. According to the findings of this study, dividend policy has a negative and considerable impact on business value. Firm value is positively affected by liquidity and firm size, but profitability has an insignificantly beneficial effect on firm value.

The aim of Setiadharma & Machali, (2017)'s research is to examine the direct and indirect effects of asset structure and firm size on firm value. This study's samples are thirty-four property and real estate enterprises that were listed on the Indonesia Stock Exchange between 2010 and 2014. The findings of this study show that (1) asset structure has a direct effect on firm value, (2) capital structure has no indirect effect on firm value with capital structure as an intervening variable, (3) firm size has no direct effect on firm value, and (4) capital structure has no indirect effect on firm value, and intervening variable.

The study conducted by Ibrahim & Isiaka, (2020) examines the impact of financial leverage on firm value using secondary data from 18 Nigerian companies. The study uses Tobin's Q ratio

as an independent variable and controls variables like Total Asset, Return on Asset, and years in operation. The study uses estimation techniques and a pairwise correlation matrix to determine the degree of causality. The results show a significant negative effect of financial leverage on firm value, with no significant linear relationship between leverage and firm value.

In Dutta *et al.*, (2018), the researchers look into the effects of financial leverage on firm value. Based on purposive sampling, the study selects 31 NSE listed companies as a sample size from six different sectors, namely, consumer goods sector, energy sector, industrial manufacturing sector, automobile sector, pharmaceutical sector, and financial services sector. The study spans ten years and employs uniformly organised data organised by fiscal year. Tobin's Q measures firm value as a dependent variable, whereas financial leverage is an independent variable. To determine the impact of financial leverage on firm value, the study employs the Fixed Effect (within) Regression Model as suggested by the Hausman Test. After controlling for the variable firm size, the study's findings show that there is a significant negative relationship between degree of financial leverage and firm value.

Nguyen & Dinh Vu, (2017) explore the effect of stock liquidity on firm value in the context of Vietnam. The study uses secondary data from the financial records of the 30 selected firms listed on Ho Chi Minh Stock Exchange (HOSE) for the periods of 2016 to 2016. Firm value as the dependent variable is proxied by Tobin's Q and the independent variable, stock liquidity is measured by turnover and Amihud Illiquidity. Based on a sample of largest firms in Vietnam from 2012 to 2016 and panel regression models, the study reports that there is no relationship between Tobin's Q and turnover volume as a proxy of stock liquidity. However, there exists a significant relationship between Tobin's Q and Amihud Illiquidity as confirmed from the random effect regression model.

2.4 Conceptual Framework

The conceptual framework is employed to evaluate the research's findings. Based on Figure 2.1 below, it was determined how the dividend policy affected the performance and value of companies listed on the Ghana Stock Exchange. The dependent variables in the study are accounting performance measures such as Tobin's Q, Return on Assets (ROA), and Return on Equity (ROE). Dividend payout ratio and dividend per yield, which serve as proxies for dividend policy, are among the explanatory variables. Other control variables that the dividend policy does not account for but which could impact a firm's value and performance are included in the study. As depicted in Figure 2.1, the control variables are firm size, leverage, and liquidity.



Dependent Variable
KNUST

CHAPTER THREE

METHODOLOGY

3.0 Introduction

This chapter describes the research methodology used in this study. Specifically, a detailed description of the study design, sample size and sampling procedure and research instruments that the study uses as well as data collection procedure and analyses

3.1 Study Design

The study is a descriptive study that uses quantitative approach involving panel estimation and logistic regression technique. Variables and correlations are the main concepts in quantitative research. Consequently, a quantitative approach is selected for this study. A significant number of data is collected and tested for the study, which is the first of several important justifications for the choice of study design. Secondly, because correlations and regression analysis is employed in the study, which requires a substantial amount of comparable data, quantitative approach of study design is the best choice for the research. Finally, after all the test and analysis is done, the quantitative research approach gives the researcher absolute values to make interpretations and come up with empirical evidence to make conclusions.

3.2 Study Population and Sample Size

3.2.1 Population

The study involves all firms listed on the Ghana Stock. According to the website of the Ghana Stock Exchange, there are now 39 firms listed on the exchange, which represent some of the key industries in the Ghanaian economy. Firms in industries like finance, pharmaceuticals, manufacturing, mining, trading, and breweries are among them.

3.2.2 Sampling Technique and Sample Size

The study uses purposive sampling technique which sets paying of dividend as the criteria for firm selection. Also, it selects firms that have not experienced any events that can have an impact on its financial position during the study period, such as integration, mergers and acquisitions, etc.; information readily available on the movement of its shares traded during the study period in the financial market; and availability of financial records. This results in the selection of 18 firms that meet the requirements for this study's selection.

3.3 Data Collection Technique

The study uses secondary data obtained from the Ghana Stock Exchange website. Audited annual financial statement of the listed firms is source of information for the dividend policy (dividend per share) and financial indicators of performance and firm value of the listed firms on the Ghana Stock Exchange (https://gse.com.gh/financial-statements/). The researcher uses panel data constructions from the financial records of firms listed in Ghana for a period of 5 years, from 2018 – 2022. The study collects data from 2018 because the financial sector of the country experienced a major clean-up which affected majority of the financial institutions in Ghana. Some of these institutions are listed on the Ghana Stock Exchange and therefore the study collects data from 2018 to eliminate firms that have undergone mergers and acquisitions. The sample is checked for duplicated data.

3.4 Study Variables

3.4.1 Dependent Variables

Performance and value of firms are used as the dependent variables of the study. Return on Equity and Return on Assets are used as the indicators of performance in separate models and Tobin's Q serves as proxy for the value of the listed firms. The return on equity is measured as ratio of net profit after tax to total equity capital and the return on assets is also measured as the ratio of net profit after tax to total current assets. Tobin's Q is calculated as the ratio of the firm's total market value to its total book value.

3.4.2 Independent Variables

Dividend yield and dividend pay-out ratio are used to evaluate the dividend policy independent variable. Dividend yield is the sum of a company's annual dividends paid to its shareholders on their investments. It also demonstrates a company's share price appeal as an investment. It is calculated by dividing the current market value of the share by: Dividend Yield equals dividends per share divided by market stock price. The second method for evaluating dividend policy is the dividend pay-out ratio, which compares a company's dividend payments to its earnings.

3.4.3 Control Variables

Firm size, leverage, and liquidity are employed as control variables in the current study. In this study, the logarithm of the total assets of the firm serves as proxy for firm size; the larger the company's total assets, the bigger the firm. The total assets log is used to calculate company size. Because large companies are able to pay out greater dividends, the log of total assets is

employed to lessen the significant disparity between the size of the too-large company and the market average (Phornlaphatrachakorn & Na-Kalasindhu, 2020). The study also uses liquidity as a control variable and measures the liquidity of a firm as the ratio of its current assets to its current liabilities. Leverage ratios represent the amount of external funding from short- and long-term borrowing. Total debts divided by total assets serves as the benchmark for this ratio.

Variable	Code	Calculation/Definition	Source	Expected Sign
Dependent Var	iables			
Return on Assets	ROA	The ratio of net profit after tax to total current assets	Annual report	
Return on Equity	ROE	The ratio of net profit after tax to total equity capital	Annual report	
Tobin's Q	TQ	The ratio of Total Market Value of Firm/Total Book Value of Firm	Annual report	
Independent Variables			1	-
Dividend per yield	DY	The ratio of dividends per share to stock price.	Annual report	7+
Dividend payout	DP	The ratio dividends paid to earnings.	Annual report	+
Control Variab	oles	Cir 1		
Size	S	Log of total assets	Annual report	+
Leverage (LEV)	LEV	The ratio of total liabilities to total assets	Annual report	
Liquidity (LIQ)	LIQ	Current assets over current liabilities	Annual report	No.

Table 3.1: Definition and source the operational variables of the study

3.5 Estimation Technique

20

A panel data regression analysis is used in the study. This is due to the fact that the data set contains observations of multiple variables over multiple time periods. This panel data incorporates both time series and cross-sectional information. It gives the researcher the

BADY

freedom to model differences in behaviour across individual firms. It is also appropriate for this study due to its ability to account for heterogeneity issues or individual effects in crosssectional data and provide more informative results (Bell *et al.*, 2019). The panel regression equation is different from a regular time-series or cross-section regression by the double subscript attached to each variable. The general form of the panel data model is specified as:

$$\mathbf{y}_i = \beta_0 + \beta \mathbf{X}_{i,t} + \boldsymbol{\varepsilon}_{i,t} \quad (1)$$

The subscript "*i*" represents the cross-sectional dimension, and the subscript "*t*" represents the timeseries dimension. The model's dependent variable, represented by the left-hand variable "y", is the performance of banks listed on the Ghana Stock Exchange. The "x" contains the estimation model, which is assumed to be constant over time t and unique to the individual cross-sectional unit "*i*".

Pearson's Correlation is used in this study find out the relationship between different variables. This is done by using the econometric model:

$$= \frac{n \sum x_{i} y_{i} - (\sum x_{i}) (\sum y_{i})}{\sqrt{[n \sum x_{i}^{2} - (\sum x)^{2}][n \sum y_{i}^{2} - (\sum y)^{2}]}}$$
(2)

For the correlation between any two variables x and y

3.6 Model Diagnostics

Several inferential statistics are used in the study. All units for the study period are considered in the cross-section data. As a result, because the data in this study is made up of cross-section and time-series data, it contains some information about all of the selected eighteen listed firms on the GSE sampled for the five years. This explains why the study uses the balanced panel estimation method. In all cases, the panel data model considers the heterogeneity of the individual group variables and thus provides the effect of the factors with less collinearity within the variables with a higher grade of choice and efficacy. For each of the dependent variable in this study, the panel model proposed is:

ROE =
$$\alpha_0 + \alpha_1 DY + \alpha_2 DP + \alpha_3 SIZE + \alpha_4 LIQ + \alpha_5 LEV + \varepsilon$$
 (3)
ROA = $\beta_0 + \beta_1 DY + \beta_2 DP + \beta_3 SIZE + \beta_4 LIQ + \beta_5 LEV + \varepsilon$ (4)
TQ = $\lambda_0 + \lambda_1 DY + \lambda_2 DP + \lambda_3 SIZE + \lambda_4 LIQ + \lambda_5 LEV + \varepsilon$ (5)
Where; ε = the stochastic error term
ROE = Return on Equity,
ROA = Return on Assets,
TQ = Tobin's Q,
DY = Dividend Yield,
DP = Dividend pay-out ratio,
SIZE = Firm size,
LIQ = Liquidity,
LEV = Leverage

One of the assumptions of the classical linear regression model is that the explanatory variables in the regression model are not multicollinear. The statistical tests used in the study to detect the presence of multicollinearity are the Variance Inflation Factor (VIF) and tolerance. The Variance Inflation Factor is calculated as follows:

$$VIF(x_i) = \frac{1}{1 - R^2} \tag{6}$$

The R^2 is the coefficient of the determination of the results of the regression of x_i against the explanatory variables. A high VIF means that the variable x_i is collinear with the other explanatory variables. VIF value of a variable that is more than 10 is deemed too be highly collinear as a rule of thumb (Lavery *et al.*, 2019). However, some authors use a more conservative threshold value of 30 (Alauddin & Nghiem, 2010; Jauk *et al.*, 2013).

The measure of tolerance is given by:

$$TOL(x_i) = \frac{1}{VIF(x_i)}$$
(7)

If the tolerance value equals 1, then x_i is not correlated with the other explanatory variables. If the tolerance value equals 0, then x_i is said to be perfectly correlated with the other explanatory variables.

3.6.2 Normality Test

This can be done graphically or mathematically, and it can be verified through visual inspection or statistical analysis. Given this condition, Alpu & Yuksek, (2016) argue that a study satisfies the multivariate normality requirement when all of its variables reach univariate normality. In order to visually determine whether the sample originates from a normal distribution, a normal probability map is created. The present study uses Shapiro-Wilk W, D'Agostino-Pearson, and Kolmogorov-Smirnov to test for the normal distribution of the panel data.

3.7 Data Analysis and Presentation

The quantitative data analysis method is used to collect and analyse the data. To assist in the analysis and evaluation of the data, financial analytical tools and techniques involving the statement of financial position and income statement items financial ratios such as liquidity, leverage, and return on asset and equity ratios are calculated using Microsoft Excel version 2019. The data is analysed in accordance with the study's primary objectives. Pearson To determine the relationship between dividend policy and the performance and value of firms listed on the Ghana Stock Exchange, correlation analysis and panel data multiple regression analysis are used. The study employs GraphPad Prism 8.0 as the primary statistical tool for analysis because it allows the researcher to conduct a thorough analysis of the data collected and presents the results in a colourful and easy-to-understand format. The findings are presented in tables and figures, and they are interpreted.



CHAPTER FOUR RESULTS AND DISCUSSION

4.0 Introduction

This chapter covers the analysis of the data collected and the interpretation of the results of the study. It provides details of descriptive and inferential statistics the study employs to examine the impact of dividend policy on firm performance and firm value.

4.1 Results

4.1.1 Descriptive Statistics

The study reports a mean of 0.995 ± 1.215 for ROA, while the mean ROE and Tobin's Q for the listed firms is 0.196 ± 0.107 and 0.315 ± 1.189 respectively. Table 4.1 gives a detailed description of the descriptive statistics of the measures of dividend policy and other control variables the study employs from 2018 to 2021. This study records a mean dividend pay-out and a dividend yield of 0.459 ± 1.233 and 0.503 ± 0.759 respectively. Also, the mean for Liquidity in this study is 2.167 ± 3.009 , while the study records 0.960 ± 1.510 as the mean for Leverage, and the mean size of the listed firms is 14.600 ± 2.060 .

	J	the second se			
Variable	N	Minimum N	Maximum	Mean	SD
Return on Assets (ROA)	60	-0.003	4.730	0.995	1.215
Return on Equity (ROE)	60	-0.003	0.517	0.196	0.107
Tobin's Q (TQ)	60	-0.001	5.539	0.315	1.189
Dividend pay-out (DP)	60	0.000	8.766	0.459	1.233

Tał	ole	4.1:	Descriptive	e statistics	of stud	y variables
-----	-----	------	-------------	--------------	---------	-------------

Dividend yield (DY)	60	0.000	2.950	0.503	0.759
Liquidity (LIQ)	60	0.099	15.000	2.167	3.009
Leverage (LEV)	60	0.026	9.081	0.960	1.510
Size (SIZE)	60	8.801	17.130	14.600	2.060

N= Number of observations, SD: Standard Deviation

Source: Researcher's Construct, 2023

4.1.2 Test for Normality

The study uses the Kolmogorov-Smirov test to determine whether the data set follows the Gaussian distribution. Table 4.2 gives a detailed description of the Kolmogorov-Smirov test of normality for all the study variables, which reports p-values less than the 0.05 alpha value. This implies that all the study variables do not follow the Gaussian normal distribution and the alternative hypothesis that the data do not follow the normal distribution should be accepted.

Variable	n	Statistic	df	p-value
Return on Assets (ROA)	60	0.213	59	< 0.001
Return on Equity (ROE)	60	0.119	59	0.035
Tobin's Q (TQ)	60	0.501	59	< 0.001
Dividend pay-out (DP)	60	0.355	59	< 0.001
Dividend yield (DY)	60	0.314	59	< 0.001
Liquidity (LIQ)	60	0.316	59	< 0.001
Leverage (LEV)	60	0.411	59	< 0.001
Size (SIZE)	60	0.130	59	0.013

Table 4.2: Kolmogorov-Smirov test of normality

N= number of observations, df = degree of freedom

4.1.3 Multicollinearity

Table 4.3 gives a detailed description of the Value Inflation Factor (VIF) and tolerance test for multicollinearity between the study dependent variables and the independent variables in the three different models the study uses to establish the impact of dividend policy on firm performance and value. In model 1, all the explanatory variables have VIF between 1.084 and

BAD

Source: Researcher's Construct, 2023

1.191 as well as tolerance values above 0. Nonetheless, dividend pay-out (DP) has a higher tolerance level in model 1 in comparison to the other explanatory variables. In model 2, the highest VIF value (2.546) is observed in the SIZE of listed firms and the lowest in Leverage (LEV). The tolerance values in model 2 with Return on Equity as the dependent variable is well below 1 and above 0. Model 3 also has VIF and tolerance values within acceptable ranges and this suggests that there is no collinearity between the variables and makes the subsequent regression analysis unbiased.

 Table 4.3: Value inflation factor (VIF) and tolerance values for the explanatory variables

Variable	Model 1		Mo	odel 2	Model 3		
	VIF	Tolerance	VIF	Tolerance	VIF	Tolerance	
DP	1.084	0.923	1.233	0.811	1.080	0.926	
DPY	1.097	0.912	1.227	0.815	1.133	0.883	
LIQ	1.191	0.840	1.348	0.742	1.216	0.822	
LEV	1.160	0.862	1.056	0.947	1.295	0.772	
SIZE	1.150	0.870	2.546	0.393	1.373	0.728	

Dependent variable: Model 1 – ROA; Model 2 – ROE; Model 3 – TQ

Source: Researchers construct, 2023

As a rule of thumb, a correlation of 0.800 or more between independent variables suggests collinearity between the variables. In Table 4.4, all the correlation coefficients are well below 0.8 which indicates there is no form of multicollinearity. Nonetheless, there is a significant association between ROA and Liquidity, Leverage, Tobin's Q, and ROE. Similarly, Firm size has a significant association with Tobin's Q, Dividend yield, and Leverage. Table 4.4 gives a detailed description of the correlation matrix of the study variables.

Table 4.4: C	Table 4.4: Correlation matrix of study variables								
Variable	ROA	ROE	TQ	DP	DPY	LIQ	LEV	SIZE	
ROA	1.000								

ROE	0.325	1.000						
TQ	-0.387	0.047	1.000					
DP	0.245	-0.013	0.030	1.000				
DPY	0.109	0.239	-0.271	0.323	1.000			
LIQ	-0.521	-0.150	0.183	0.058	0.030	1.000		
LEV	0.266	-0.097	-0.461	-0.019	0.181	-0.230	1.000	
SIZE	0.420	0.179	-0.367	0.199	0.230	-0.209	0.555	1.000

Boldened values are statistically significant at p<0.05

4.1.4 Model Summary and Test for Autocorrelation

The study employs the Durbin-Watson test for serial correlation to evaluate the data for autocorrelation which is the similarity of a time series data over a successive time interval and can lead to underestimation of standard error. This consequently masks the predictors as falsely significant. Table 4.5 gives a detailed description of the Durbin-Watson test for autocorrelation for the study variables across the three models the study examines. All three models have positive autocorrelation with values in the normal range of 1.5 to 2.5.

In Table 4.5, the R² value for Model 1 is 0.263 which indicates that 26.3% of the variations in the dependent variable, return on asset (ROA) as a measure of firm performance is explained by the independent and control variables. The remaining 73.7% is explained by other factors and the error term in the model. Also, in Model 2, the independent and control variables explain 34.3% (R² = 0.343) of the variation in the dependent variable, return on equity which also serves as an indicator of firm performance in this study. The error term and other variables explain the remaining 65.7% of the variation in the dependent variable, ROE. In Model 3, the independent and control variables explain 64.9% of the variation in the dependent variable Tobin's Q (TQ), which serves as an indicator for firm value in the study. The error term and other factors explain the remaining 35.1% of the variation in the dependent variable, Tobin's Q.

Source: Researcher's Construct, 2023

Model	R	R2	Adjusted R ²	Standard Error of Estimate	Durbin- Watson
1	0.513	0.263	0.195	1.043	1.519
2	0.585	0.343	0.228	0.209	1.509
3	0.806	0.649	0.607	0.713	1.61

Table 4.5: Durbin-Watson test for autocorrelation

Predictors (constant): Dividend pay-out, Dividend yield, Liquidity, Leverage, and size Dependent variable(s): 1. Return on Assets, 2. Return on Equity, 3. Tobin's Q Source:

Researcher's construct, 2023

4.1.5 Analysis of Variance

As Table 4.6 shows, the statistic value is 6.838 which is significant at a 5% significance level with a p-value <0.001. These ANOVA findings indicate that the regression model is significant and a good predictor of the relationship between dividend policy and firm performance using ROA as a proxy of performance.

Model: 1	Sum of Squar	es df	Mean Squ	are F	p-value
Regression	6.659	5	1.332	6.838	< 0.001
Residual	8.569	54	0.195		
Total	15.23	59	1		

Dependent Variable: Return on Asset (ROA)

Predictors: (Constant), Dividend pay-out, Dividend yield, Liquidity, Leverage, and Firm size.

Source: Researcher's construct, 2023

Table 4.7 depicts that the statistic value of 1.694 is not significant at the alpha value of 0.05 and this finding from the ANOVA test indicates that there is no significant association between dividend policy and firm performance using return on equity as a proxy for firm performance. Therefore, the model may not predict the true nature of the relationship between dividend policy and firm performance.

Sum of Squ	ares df	Mean Squ	are F	p-value
0.434	5	0.087	1.694	0.156
2.254	54	0.051		
2.688	59	100 100	_	
	0.434 2.254 2.688	Sum of squares df 0.434 5 2.254 54 2.688 59	Sum of squares df Mean squares df 0.434 5 0.087 2.254 54 0.051 2.688 59	Sum of squares df Mean square F 0.434 5 0.087 1.694 2.254 54 0.051 2.688 59

Table 4.7: Analysis of Variance for the model with ROE as the dependent variable

Dependent Variable: Return on Equity (ROE)

Predictors: (Constant), Dividend pay-out, Dividend yield, Liquidity, Leverage, and Firm size.

Source: Researcher's construct, 2023

In Model 3, as Table 4.8 shows, the statistic 15.200 is significant at a 5% significance level with a p-value of <0.001. These findings from the Analysis of Variance indicate that the regression model is significant and a good predictor of the relationship between dividend policy and firm value using Tobin's Q (TQ) as a proxy of firm value.

Table 4.8: Analysis of Variance for the model with TQ as the dependent variable

Model: 3	Sum of Squa	ares df	Mean Squ	are F	p-value
Regression	43.340	5	<mark>8.</mark> 668	15.200	< 0.001
Residual	23.380	54	0.570	227	
Total	66.720	59	X-155	North Contraction	

Dependent Variable: Return on Equity (ROE)

Predictors: (Constant), Dividend pay-out, Dividend yield, Liquidity, Leverage, and Firm size.

Source: Researcher's construct, 2023

4.1.6 Regression Analysis of Study Variables

The regression analysis in Model 1 shows that there is a significant positive association between return on assets (ROA) as an indicator of performance and dividend pay-out (DP) which serves as an indicator for dividend policy (β =0.379, t = 1.164, p=0.022). Also, ROA has a significant and strong positive association with liquidity (β = 0.643, t = 3.884, p<0.001) and firm size (β = 2.146, t = 2.181, p = 0.035) as Table 4.9 depicts. Furthermore, ROA has an insignificant but

positive association with dividend yield ($\beta = 0.133$, t = 1.612, p = 0.114) and leverage ($\beta = 0.287$, t = 1.661, p = 0.104) in firms listed in Ghana.

In Model 2, return on equity (ROE) has a significant and positive association with dividend pay-out ($\beta = 0.351$, t = 0.534, p = 0.047) but has an insignificant and negative association with dividend yield ($\beta = -0.146$, t = -1.037, p = 0.306). In Table 4.9, it is observed that ROE has an insignificant but positive association with liquidity ($\beta = 0.158$, t = 1.117, p = 0.271) and firm size ($\beta = 0.727$, t = -0.922, p = 0.362) of firms listed in Ghana. Furthermore, the study reports a negatively insignificant association between ROE and leverage ($\beta = -0.088$, t = 0.306, p = 0.306).

0.761) as Table 4.9 depicts.

In Model 3, Tobin's Q (TQ) as a measure of firm value has a significant and positively strong association with firm size ($\beta = 12.980$, t = 8.465, p<0.001) and also has a significant but negatively strong association with leverage ($\beta = -0.826$, t = 2.067, p = 0.045). In the same model, ROE has an insignificantly positive association with dividend pay-out ($\beta = 0.424$, t = 1.571, p = 0.125). As Table 4.9 depicts, ROE has an insignificantly negative association with dividend yield ($\beta = -0.266$, t = 1.736, p = 0.091) and liquidity ($\beta = -0.333$, t = 1.002, p = 0.323).

The regression results in Table 4.9 leads to the formation of the following equations:

Model 1:

ROA = 0.379 (DP) + 0.133 (DPY) + 0.643 (LIQ) - 0.289 (LEV) + 2.146 (SIZE) - 2.767 Model

2:

ROE = 0.351 (DP) - 0.146 (DPY) + 0.158 (LIQ) - 0.088 (LEV) + 0.727 (SIZE) - 0.991 Model3:

$$TQ = 0.424 (DP) - 0.266 (DPY) - 0.333 (LIQ) - 0.826 (LEV) + 12.980 - 10.470$$



Variable	Model 1				Model 2				Model 3			
	β	SE	t	p-value	β	SE	t	p-value	β	SE	t	p-value
Intercept	-2.767	1.164	2.377	0.022	-0.991	0.713	-1.39	0.015	- 10.470	2.213	4.729	< 0.001
DP	0.379	0.144	2.631	0.012	0.351	0.657	0.534	0.047	0.424	0.270	1.571	0.125
DPY	0.133	0.082	1.612	0.114	-0.146	0.057	-1.037	0.306	-0.266	0.153	1.736	0.091
LIQ	0.643	0.166	3.884	<0.001	0.158	0.043	1.117	0.271	-0.333	0.332	1.002	0.323
LEV	0.287	0.173	1.661	0.104	-0.088	0.094	0.306	0.761	-0.826	0.400	2.067	0.045
SIZE	2.146	0.984	2.181	0.035	0.727	0.505	-0.922	0.362	12.980	1.534	8.465	< 0.001

Table 4.9: Relationship between dividend policy, liquidity, leverage and size, and the performance and value of listed firms

Dependent Variable(s): Model 1 – Return on Assets; Model 2 – Return on Equity; Model 3 – Tobin's Q

THE AP 3 W 3 SANE

β: Coefficient of Regression, SE: Standard Error of Coefficient, DP: Dividend pay-out, DPY: Dividend yield, LIQ: Liquidity, LEV: Leverage, SIZE: Firm Size

BADWS

NO

Source: Researcher's construct, 2023



4.2 Discussion

This study examines the connection between dividend policy, as measured by dividend payout and dividend yield, and firm performance, as measured by return on asset and return on equity, as well as the connection between dividend policy and firm value, as measured by Tobin's Q. Based on the data in Table 4.9, it appears that dividend payout and dividend yield positively affect ROA and ROE for publicly traded companies in Ghana. There was also statistical evidence that dividend distribution had an effect. According to the data, dividend yield has a positive and statistically significant effect on ROA but a small and inconsequential effect on ROE (Table 4.9). This research's results are in line with those of a paper from Vietnam by Nguyen et al., (2021) where the researchers indicate in their report that dividend pay-out has a positive and significant impact on ROA. Similarly, Kanakriyah, (2020) and Habumugisha & Mulyungi, (2018) also report a positive association between dividend pay-out and firm performance (ROA and ROE) which are consistent with the discoveries of this research. However, the finding of this study varies from that of Narang, (2018) who concludes that there is no significant association between return on asset and dividend pay-out, which is similar to the report of Ugwu et al., (2020) where the authors indicate that dividend pay-out has a negligible relationship with ROE. The findings of the study indicate that an increase in dividend pay-out leads to an increase in the financial performance (ROA and ROE) of listed firms and this also supports the Bird-In-Hand Theory. ROA offers an idea of how effective management is at utilizing its assets to create profits and indicates how efficient management is (Puspitasari et al., 2021). If the company is doing well financially, it can decide how much of a dividend to pay out in accordance with the hopes of its shareholders without jeopardizing its own ability to thrive. The dividend pay-out ratio is a useful measure of both the company's current financial health and the confidence of its shareholders in the company's future prospects.

Researchers found that listed companies whose liquidity was high also performed better. Also, Table 4.9 shows that while liquidity has a favorable impact on ROE for shareholders, this effect is too small to be statistically significant. Kong et al., (2019) find similar results, demonstrating that liquidity significantly correlates with a firm's financial performance as assessed by ROA but has no bearing on the firm's financial performance as evaluated by ROE. The results of this study are consistent with those of another Nigerian study by the same author, which also found that increased liquidity improved financial performance (Horsfall, 2022), In addition, the findings of this study are comparable with those of another Pakistani study (Samo & Murad, 2019), that reports liquidity has a negative effect on return on equity (ROE) and a positive effect on return on assets (ROA). Li et al., (2020) find that the opposite is true, arguing that increased liquidity significantly reduces ROE. One reason why liquidity has a negative impact on return on equity is because public companies tend to overinvest in current assets like inventory that are difficult to turn over (Li et al., 2020). Stockpiles have a detrimental impact on a company's performance measured by return on equity because of the significant storage and other carrying costs they incur over time (Cappa et al., 2021). One possible explanation for the positive correlation between ROA and liquidity is that an increase in ROA favorably affects both current assets and net working capital. Both the composition of financing sources and the composition of assets have an impact on a company's liquidity and profitability. From a financial liquidity standpoint, the most important issue is the percentage of short-term liabilities in capital financing projects. Once the level of financial liquidity falls below a predetermined minimal level, the level of profitability also begins to fall. If a company has insufficient cash on hand, it will be unable to meet its debt obligations on time.

Return on assets (ROA) is positively affected by leverage, whereas return on equity (ROE) is negatively affected by it. Furthermore, Table 4.9 shows that there is no statistically significant relationship between leverage and financial performance. However, Bunyaminu *et al.*, (2021)

find that leverage significantly reduces banks' earnings regardless of the proxy for profitability used in their study. There is some inconsistency between the conclusions of this study and those of Gathara *et al.*, (2019), who suggest that leverage had a significant favorable influence on the financial performance of selected companies listed at NSE, Kenya. Financial leverage, as reported by Iqbal & Usman, (2018) and Das *et al.*, (2021), has a considerable negative impact on a firm's return on equity and a significant favorable impact on its return on assets. Higher levels of debt have a negative impact on a company's performance and lower stock value (Kalantonis *et al.*, 2021). This is because enterprises are more likely to borrow money and make their contractual payments on time if they expect a higher return on equity. This research lends credence to the Pecking Order Hypothesis. However, return on asset (ROA) is positively affected by leverage in this study, suggesting that the greater a company's asset base, the greater its ability to earn income without increasing its debt load.

Return on assets is shown to be positively strong and substantial regardless of the size of the firm, but return on equity is found to be insignificantly strong (Table 4.9). This finding is consistent with the results of a similar study conducted in Pakistan (Aydın-Unal *et al.*, 2017), which found that a company's size has a considerable positive influence on the ROA. Aduralere-Opeyemi, (2019), Eyigege, (2018) and Olawale *et al.*, (2017), in contrast to the results of this study, report a positive but insignificant effect of firm size on return on equity but a negative but insignificant effect of firm size on return on equity but a negative but insignificant effect of firm size has no bearing on a company's success. Similar findings were found in a study conducted in Ghana by Musah & Kong, (2019), which demonstrated a positive relationship between firm size and return on investment (ROI). This study did discover a small negative correlation between business size and return on equity, but the effect was not statistically significant. There is a positive correlation between the size of a company and its performance, as measured by ROA, which

suggests the company is expanding; a higher ROA corresponds to a larger business, therefore the more assets and revenue a company has, the bigger it is.

Several studies have investigated how dividend policy affects the value of businesses. Firm value is affected in a complex way by dividend policy, according to our analysis (Table 4.9). When dividends are used as a proxy for dividend policy, Tobin's Q (a measure of firm value) rises slightly but not noticeably. In addition, the dividend yield has a little negative impact on the market value of publicly traded companies in Ghana. Results from this study differ from those reported by Dang et al., (2021), who claim that corporations with a larger dividend payout policy tend to be more valuable. In addition, the findings of this study conflict with those of a study conducted in Nigeria (Nympha et al., 2021), which found that dividend pay-out has a negligible negative effect on company value while dividend yield has a sizable negative effect on firm value. However, other Indonesian research (Nurokhmah et al., 2023; Sondakh, 2019) demonstrate a favourable influence of dividend pay-out on firm value, which is consistent with the results of our study. In addition, Bezawada & Tati, (2017) discover that dividend yield has a negative effect on company value, which is consistent with the results of the present investigation. This research shows that dividend policy has little effect on firm value for firms that either do not pay dividends at all or pay them at a low rate. Regular dividend payments demonstrate to investors that the company's leadership is confident in its ability to continue generating profits. This boosts the company's allure to investors, which in turn raises the stock price and improves the company's value.

Table 4.9 shows that the influence of liquidity on firm value is small and statistically negligible. Batten & Vo, (2019) also show a negative effect of liquidity on business value, which is consistent with the results of this study. Liquidity also has a little negative impact on business value, according to an Indonesian study (Zuhroh, 2019). On the other hand, multiple researchs (Jihadi *et al.*, 2021; Putra & Astika, 2019; Rafid *et al.*, 2019; Sondakh, 2019) also find that liquidity has a considerable positive influence on the value of a corporation. The higher the liquidity ratio of a corporation, the greater the proportion of its liabilities that are covered by current assets, which in turn increases investor confidence and the firm's value. Even though this study found a negative influence, it's likely that investors don't pay much attention to a company's liquidity because the current ratio simply reflects the company's capacity to meet its short-term obligations with its current assets. The stock market responds favourably to a firm's capacity to satisfy its short-term obligations because, according to the signal theory, this suggests the company will be able to sustain its success, which in turn should increase its value.

However, this research shows that increasing debt levels considerably reduces a company's worth. As can be seen in Table 4.9, there is a negative correlation between leverage and business value, corroborated by the research of Ibrahim & Isiaka, (2020). Another study conducted in Nigeria (Akani & Ifechi, 2017) confirms the detrimental impact of financial leverage on business value found in this one. The findings of this study are consistent with those of other studies conducted in Vietnam (Luu, 2021; Mai, 2020; Nguyen-Trong & Nguyen, 2020), all of which reveal a strong negative effect of leverage on business value. A number of other research, however (Aprilyani *et al.*, 2021; Jihadi *et al.*, 2021; Nurokhmah *et al.*, 2023; Santosa, 2020), indicate a strong beneficial effect of leverage on business value. Consistent with the "Pecking-Order Theory," this study finds that a company's value and attractiveness to investors both peak at high levels of leverage or debt to equity.

Value of listed firms is positively correlated with firm size, as measured by the natural logarithm of total assets (Table 4.9). Researchers in Vietnam found that larger firms have a more positive effect on company value, and that this link is statistically significant (T. D. Dang & Do, 2021; Nguyen-Trong & Nguyen, 2020). In addition, Sondakh (2019) found similar results in

Indonesia, reporting that larger firms have a greater favorable impact on company value. This study's conclusions are at odds with those of other researchers, such as Aprilyani *et al.* (2021), who found only a little negative influence of firm size on firm value. Firm size has a large positive effect on firm value, according to Lumapow & Tumiwa, (2017), but Nurokhmah *et al.*, (2023) report an insignificant positive effect on company value. A different study (Luu, 2021) finds the opposite, that larger firms have a negative influence on smaller ones. The size of a firm has a direct bearing on its value, as larger companies are more likely to attract investors who will have an effect on the company's worth in the future.



CHAPTER FIVE

SUMMARY OF FINDINGS, CONCLUSION AND RECOMMENDATIONS

5.0 Introduction

This chapter is divided into three sections. The first section covers the summary of the main findings of the study that accomplish the objectives of the study. The second section covers the conclusions the researcher draws from the findings of the study and the final section covers relevant recommendations based on the findings of the study.

5.1 Summary of Findings

From the analysis of the data the study uses, dividend policy has a substantial impact on financial performance. except when dividend policy is proxied by dividend yield, which shows both insignificant positive and negative effects of ROA and ROE respectively.

The study reveals that liquidity and firm size have a significant positive impact on firm performance (ROA) but shows an insignificant positive effect on ROE as a measure of financial performance. Leverage has an impact on ROA and ROE in both positive and negative ways respectively.

Dividend policy affects firm value both positively and negatively when dividend pay-out and dividend yield serve as a proxy for dividend policy respectively.

With regards to the size of the listed firm, it shows a substantial positive impact on firm value, while leverage exerts a considerable negative influence on the value of the listed firms. Also, liquidity has an insignificant impact on the value of listed firms in Ghana.

5.2 Conclusion

This study examines the impact of dividend policy on the financial performance and value of listed firms in Ghana. The study employs purposive sampling technique to select 15 firms listed on the Ghana Stock Exchange and uses secondary data from the audited financial statement of the firms that meet the inclusion criteria of the study. The study uses an ordinary least square regression model to establish the relationship between dividend policy and financial performance and firm value and simultaneously examines the effect of other control variables (leverage, liquidity, and firm size) on financial performance and firm value. From the findings of the study, the researcher concludes that dividend policy has a significant impact of the financial performance of listed firms in Ghana, thus firms perform well when dividend shareholders receive increases. More so, the findings also indicate that larger firms tend to perform better than firms of small sizes. Also, the researcher concludes that firms with higher levels of liquidity tend to outperform firms with lower levels of liquidity. Furthermore, firms with a higher degree of leverage, thus using debt to finance their operations, perform better financially. Notwithstanding this, on the basis of the findings of this study, the researcher concludes that firms with higher degree of leverage experience a significant decrease in the valuation of the firm. However, liquidity does not have any significant effect on the value of listed firms in Ghana.

5.3 Recommendations

Based on the current study's findings, the researcher suggests the following. Financial managers need to carefully evaluate the level of leverage that is right for their business, weighing the possible advantages of debt financing operations against the impact on the firm's valuation. For the firm to avoid taking on excessive risk and to maintain its overall financial stability, a careful approach to controlling leverage is essential. Financial managers should evaluate the company's financial standing and, if feasible, explore raising dividend payments. Since dividend policy has a substantial impact on the financial performance of listed corporations in Ghana, this could aid the company attract investors and possibly improve its financial performance.

Financial managers should work to increase their companies' size and growth potential. This can be accomplished by tactics including acquisitions, market expansion, and the creation of new product lines.

Despite the study's conclusion that value is not significantly affected by liquidity. of Ghana's listed companies, financial managers should nonetheless take into account other aspects that affect firm valuation. When making strategic financial decisions, other factors including profitability, growth prospects, market conditions, and industry-specific dynamics should be taken into consideration.

Future studies should focus on other variables when evaluating the impact of the dividend policy (earnings per share, dividend per share) on financial performance (Return on Investment), and firm value, since the results of the study indicate the absence of a statistically significant impact of the dividend yield on firm performance and value, and to also apply more control variables such as board structure, governance techniques, and internal or external auditing, and business risk to enhance and make the results tougher and stronger.

BADW

51

SAP J W J SANE

REFERENCES

- Abor, J., & Amidu, M. (2006). Determinants of dividend payout ratios in Ghana. *Journal of Risk Finance*, 7, 136–145. https://doi.org/10.1108/15265940610648580
- Adam, K., Marcet, A., & Beutel, J. (2017). Stock price booms and expected capital gains. *American Economic Review*, 107(8), 2352–2408.
- Aduralere-Opeyemi, O. (2019). The Impact of Firm Size on Firms Performance in Nigeria: A Comparative Study of Selected Firms in the Building Industry in Nigeria. Asian Development Policy Review, 7(1), 1–11. https://doi.org/10.18488/journal.107.2019.71.1.11
- Ahmed, F., Awais, I., & Kashif, M. (2018). Financial leverage and firms' performance:Empirical evidence from KSE-100 Index. *Etikonomi: Jurnal Ekonomi*, 17(1), 45–56.
- Akani, H. W., & Ifechi, K.-N. J. (2017). Effects of capital structure and board structure on corporate performance of selected firms in Nigeria. *Indian Journal of Finance and Banking*, 1(2), 1–16.
- Akenga, G. (2017). Effect of liquidity on financial performance of firms listed at the Nairobi Securities Exchange, Kenya. *International Journal of Science and Research*, 6(7), 279– 285.
- Akisik, O., & Gal, G. (2019). Integrated reports, external assurance and financial performance:
 An empirical analysis on North American firms. *Sustainability Accounting, Management and Policy Journal*, 11(2), 317–350. https://doi.org/10.1108/SAMPJ-02-2019-0072
- Alauddin, M., & Nghiem, H. S. (2010). Do instructional attributes pose multicollinearity problems? An empirical exploration. *Economic Analysis and Policy*, 40(3), 351–361.
- Allen, F., & Michaely, R. (2003). Payout policy. In Handbook of the Economics of Finance (1st ed., Vol. 9, pp. 337–429). Elsevier.

- Almashhadani, M., & Almashhadani, H. A. (2022). The beneficial of firm size, board size, ownership structure, and independence in developing markets' firm performance:
 Evidence from Asia. *International Journal of Business and Management Invention*, *11*(7), 88–92.
- Alpu, Ö., & Yuksek, D. (2016). Comparison of some multivariate normality tests: A simulation study. *INTERNATIONAL JOURNAL OF ADVANCED AND APPLIED SCIENCES*, 3(12), 73–85. https://doi.org/10.21833/ijaas.2016.12.011
- Amidu, M. (2007a). Determinants of capital structure of banks in Ghana: An empirical approach. *Baltic Journal of Management*, 2(1), 314–332.
- Amidu, M. (2007b). How does dividend policy affect performance of the firm on Ghana stock
 Exchange. *Investment Management and Financial Innovations*, 4(2), 103–112.
- Aprilyani, I., Widyarti, M. T. H., & Hamida, N. (2021). The effect of ERM, firm size, leverage, profitability and dividend policy on firm value (evidence from food & beverage sub sector companies listed in. Jurnal Aktual Akuntansi Keuangan Bisnis Terapan (AKUNBISNIS), 4(1), Article 1. https://doi.org/10.32497/akunbisnis.v4i1.2663
- Aryati, A. (2017). Effect of profitability and dividend policy on corporate governance and firm value: Evidence from the Indonesian manufacturing Sectors. *IOSR Journal of Business* and Management, 19(10), 41–49.
- Atidhira, A. T., & Yustina, A. I. (2017). The influence of return on asset, debt to equity ratio, earnings per share, and company size on share return in property and real estate companies. *JAAF (Journal of Applied Accounting and Finance)*, *1*(2), 128–146.

Aydın-Unal, E., Unal, Y., & Isık, O. (2017). The Effect of Firm Size on Profitability: Evidence

From Turkish Manufacturing Sector. Pressacademia, 6(4), 301–308.
 https://doi.org/10.17261/Pressacademia.2017.762
 Ayunku, P. E., & Apiri, R. T. (2020). Dividend Policy Impact on Market Value of Quoted

Commercial Banks in Nigeria (2004-2018). Saudi Journal of Business and

 Management
 Studies,
 05(03),
 181–195.

 https://doi.org/10.36348/sjbms.2020.v05i03.002
 181–195.
 181–195.

 Azaro, K., Djajanto, L., & Sari, P. A. (2020). The Influence of Financial Ratios and Firm Size on Firm Value (An Empirical Study on Manufacturing Companies Sector Consumers Goods Industry Listed in Indonesian Stock Exchange in 2013–2017). Advances in Economics, Business and Management Research, 136, 142–147.

Banerjee, B. (2015). Fundamentals of financial management. PHI Learning Pvt. Ltd.

- Bataineh, H. (2021). The impact of ownership structure on dividend policy of listed firms in Jordan. *Cogent Business & Management*, 8(1), 1863175.
- Batten, J., & Vo, X. V. (2019). Liquidity and firm value in an emerging market. *The Singapore Economic Review*, 64(02), 365–376. https://doi.org/10.1142/S0217590817470063
- Bell, A., Fairbrother, M., & Jones, K. (2019). Fixed and random effects models: Making an informed choice. *Quality & Quantity*, 53(2), 1051–1074. https://doi.org/10.1007/s11135-018-0802-x
- Bezawada, B., & Tati, R. K. (2017). Dividend Policy and Firm Valuation—A Study of Indian Electrical Equipment Manufacturing Industry. *Theoretical Economics Letters*, 7(5), Article 5. https://doi.org/10.4236/tel.2017.75083
- Bibi, N., & Amjad, S. (2017). The relationship between liquidity and firms' profitability: A case study of Karachi Stock Exchange. *Asian Journal of Finance & Accounting*, 9(1), 54–67.
- Budagaga, A. (2017). Dividend payment and its impact on the value of firms listed on Istanbul stock exchange: A residual income approach. *International Journal of Economics and Financial Issues*, 7(2), 370–376.

- Bunyaminu, A., Yakubu, I. N., & Bashiru, S. (2021). The Effect Of Financial Leverage On Profitability: An Empirical Analysis Of Recapitalized Banks In Ghana. *International Journal of Accounting & Finance Review*, 7(1), 93–102.
- Busse, J. A., Chordia, T., Jiang, L., & Tang, Y. (2021). Transaction costs, portfolio characteristics, and mutual fund performance. *Management Science*, 67(2), 1227–1248.
- Cappa, F., Oriani, R., Peruffo, E., & McCarthy, I. (2021). Big data for creating and capturing value in the digitalized environment: Unpacking the effects of volume, variety, and veracity on firm performance. *Journal of Product Innovation Management*, 38(1), 49–67.
- Dang, H. N., Vu, V. T. T., Ngo, X. T., & Hoang, H. T. V. (2021). Impact of dividend policy on corporate value: Experiment in Vietnam. *International Journal of Finance & Economics*, 26(4), 5815–5825.
- Dang, T. D., & Do, T. V. T. (2021). Does capital structure affect firm value in Vietnam. Investment Management and Financial Innovations, 18(1), 33–41.
- Das, N. C., Chowdhury, M. A. F., & Islam, Md. N. (2021). The heterogeneous impact of leverage on firm performance: Empirical evidence from Bangladesh. *South Asian Journal of Business Studies*, *11*(2), 235–252. https://doi.org/10.1108/SAJBS-04-2020-0100
- Dutta, S. R., Mukherjee, T., & Sen, S. S. (2018). Impact of financial leverage on the value of firm: Evidence from some NSE listed companies. *The BESC Journal of Commerce and Management*, 4(5), 42–52.
- Eka, H. (2018a). Corporate finance and firm value in the Indonesian manufacturing companies. Business Studies, 11(2), 113–127.
- Eka, H. (2018b). Corporate finance and firm value in the Indonesian manufacturing companies. *Business Studies*, *11*(2), 113–127.

- Emeni, F. K. (2017). The Effect of Dividend Policy on Market Value of Firms: Evidence from Nigeria. *Finance India*, *31*(1), 175–208.
- Enekwe, C. I., Agu, C. I., & Eziedo, K. N. (2014). The effect of financial leverage on financial performance: Evidence of quoted pharmaceutical companies in Nigeria. *IOSR Journal of Economics and Finance*, *5*(3), 17–25.
- Enekwe, C. I., Nweze, A. U., & Agu, C. (2015). The effect of dividend payout on performance evaluation: Evidence of quoted cement companies in Nigeria. *European Journal of Accounting, Auditing and Finance Research*, 3(11), 40–59.
- Eyigege, A. I. (2018). Influence of firm size on financial performance of deposit money banks quoted on the Nigeria Stock Exchange. *International Journal of Economics and Financial Research*, 4(9), 297–302.
- Gathara, Z. M., Kilika, J. M., & Maingi, J. N. (2019). Effect of Leverage on Financial Performance of Selected Companies Listed in the Nairobi Securities Exchange, Kenya. *Int. J. Innovative Finance and Economics Res*, 7(1), 10–33.
- Habumugisha, T., & Mulyungi, P. (2018). Effect of corporate dividend policy on performance of stock prices in Rwanda Stock Exchange. Case study of Bank of Kigali as listed on Rwanda Stock Exchange (2011-2016). *International Journal of Research in Management, Economics and Commerce*, 8(5), 183–193.
- Hansda, M., Sinha, A., & Bandopadhyay, K. (2020). Impact of Dividend Policy on Firm Value with Special Reference to Financial Crisis. 10, 158–175.
- Hasan, M., Ahmad, M. I., Rafiq, M. Y., & ur Rehman, R. (2015). Dividend payout ratio and firm's profitability: Evidence from Pakistan. *Theoretical Economics Letters*, *5*, 441–445.

Horsfall, K. A. (2022). Liquidity And Financial Performance Of Listed Non-Financial

Companies In Nigeria. *International Journal of Innovative Finance and Economics Research*, *10*(4), 111–122.

- Husain, T., & Sunardi, N. (2020). Firm's Value Prediction Based on Profitability Ratios and Dividend Policy. *Finance & Economics Review*, *2*(2), 13–26.
- Husna, A., & Satria, I. (2019). Effects of return on asset, debt to asset ratio, current ratio, firm size, and dividend payout ratio on firm value. *International Journal of Economics and Financial Issues*, 9(5), 50.
- Ibrahim, U. A., & Isiaka, A. (2020). Effect of financial leverage on firm value: Evidence from selected firms quoted on the Nigerian stock exchange. *European Journal of Business* and Management, 12(3), 124–135.
- Iqbal, U., & Usman, M. (2018). Impact of Financial Leverage on Firm Performance: Textile Composite Companies of Pakistan. *SEISENSE Journal of Management*, *1*(2), Article 2. https://doi.org/10.33215/sjom.v1i2.13
- Jabbouri, I. (2016). Determinants of corporate dividend policy in emerging markets: Evidence from MENA stock markets. *Research in International Business and Finance*, 37, 283– 298.
- Jauk, E., Benedek, M., Dunst, B., & Neubauer, A. C. (2013). The relationship between intelligence and creativity: New support for the threshold hypothesis by means of empirical breakpoint detection. *Intelligence*, 41(4), 212–221.
- Jihadi, M., VILANTIKA, E., HASHEMI, S. M., Arifin, Z., BACHTIAR, Y., & Sholichah, F. (2021). The effect of liquidity, leverage, and profitability on firm value: Empirical evidence from Indonesia. *The Journal of Asian Finance, Economics and Business*, 8(3), 423–431.
- Kadim, A., Sunardi, N., & Husain, T. (2020). The modeling firm's value based on financial ratios, intellectual capital and dividend policy. *Accounting*, 6(5), 859–870.

- Kalantonis, P., Kallandranis, C., & Sotiropoulos, M. (2021). Leverage and firm performance: New evidence on the role of economic sentiment using accounting information. *Journal* of Capital Markets Studies, 5(1), 96–107. https://doi.org/10.1108/JCMS-10-2020-0042
- Kanakriyah, R. (2020). Dividend Policy and Companies' Financial Performance. *The Journal* of Asian Finance, Economics and Business, 7(10), 531–541. https://doi.org/10.13106/jafeb.2020.vol7.no10.531
- Khan, M. K., Ali, R., & Tariq, S. (2018). Impact of Ownership Structure On Corporate Dividend Policy and Performance. *KASBIT Business Journal*, 11(1), 110–130.
- Khan, M. N., Naeem, M. U., Rizwan, M., & Salman, M. (2016). Factors affecting the firm dividend policy: An empirical evidence from textile sector of Pakistan. *International Journal of Advanced Scientific Research and Management*, 1(5), 144–149.
- Kim, T., & Kim, I. (2020). The influence of credit scores on dividend policy: Evidence from the Korean market. *The Journal of Asian Finance, Economics and Business*, 7(2), 33–42.
- Kong, Y., Musah, M., & Agyemang, A. O. (2019). Liquidity and financial performance: A correlational Analysis of Quoted Non-financial firms in Ghana. *International Journal* of Trend in Scientific Research and Development (IJTSRD), 3(5), 133–143.
- Lavery, M. R., Acharya, P., Sivo, S. A., & Xu, L. (2019). Number of predictors and multicollinearity: What are their effects on error and bias in regression? *Communications in Statistics-Simulation and Computation*, 48(1), 27–38.
- Li, K., Musah, M., Kong, Y., Adjei Mensah, I., Antwi, S. K., Bawuah, J., Donkor, M., Coffie, C. P. K., & Andrew Osei, A. (2020). Liquidity and Firms' Financial Performance Nexus: Panel Evidence From Non-Financial Firms Listed on the Ghana Stock Exchange. SAGE Open, 10(3), 2158244020950363.

https://doi.org/10.1177/2158244020950363

- Lumapow, L. S., & Tumiwa, R. A. F. (2017). The effect of dividend policy, firm size, and productivity to the firm value. *Research Journal of Finance and Accounting*, 8(22), 20– 24.
- Luu, D. H. (2021). The impact of capital structure on firm value: A case study in Vietnam. *The Journal of Asian Finance, Economics and Business*, 8(5), 287–292.
- Mai, T. G. (2020). The impact of capital structure on the value of firms in the plastic and packaging industry listed in Vietnam. *Economy and Development Journal*, 280, 59–67.
- Markonah, M., Salim, A., & Franciska, J. (2020). Effect of profitability, leverage and liquidity to the firm value. *Dinasti International Journal of Economics, Finance & Accounting*, *1*(1), 83–94. https://doi.org/10.38035/dijefa.v1i1.225
- Miller, M. H., & Modigliani, F. (1961). Dividend policy, growth, and the valuation of shares. *The Journal of Business*, *34*(4), 411–433.
- Musah, M., & Kong, Y. (2019). The relationship between size and firms' financial performance: An empirical evidence from the Ghana stock exchange (GSE). *Journal of Management Research and Analysis*, 6(1), 64–72.
- Narang, M. (2018a). Dividend policy and firm performance: A study of listed firms on national stock exchange. *International Journal of Advanced Educational Research*, 3(1), 286– 289.
- Narang, M. (2018b). Dividend policy and firm performance: A study of listed firms on national stock exchange. *International Journal of Advanced Educational Research*, *3*(1), 286–289.
- Ngo, T. G., & Dang, A. T. (2016). Financial performance, dividend payment and firm value an exploratory research on Vietnam listed firms in the food and drink industry. *VNU* JOURNAL OF ECONOMIC AND BUSINESS, 32(2).

- Nguyen, A. H., Pham, C. D., Doan, N. T., Ta, T. T., Nguyen, H. T., & Truong, T. V. (2021a). The Effect of Dividend Payment on Firm's Financial Performance: An Empirical Study of Vietnam. *Journal of Risk and Financial Management*, 14(8), Article 8. https://doi.org/10.3390/jrfm14080353
- Nguyen, A. H., Pham, C. D., Doan, N. T., Ta, T. T., Nguyen, H. T., & Truong, T. V. (2021b). The Effect of Dividend Payment on Firm's Financial Performance: An Empirical Study of Vietnam. *Journal of Risk and Financial Management*, 14(8), Article 8. https://doi.org/10.3390/jrfm14080353
- Nguyen, M. H. T., & Dinh Vu, N. H. (2017). *The Impact Of Stock Liquidity On Firm Value: Evidence From Vietnam.* 30th Australasian Finance and Banking Conference.
- Nguyen Trong, N., & Nguyen, C. T. (2020). Firm performance: The moderation impact of debt and dividend policies on overinvestment. *Journal of Asian Business and Economic Studies*, 28(1), 47–63. https://doi.org/10.1108/JABES-12-2019-0128
- Nkn, E. (2018). Determinants of Dividend Policy among Banks Listed on the Ghana Stock Exchange. Journal of Business & Financial Affairs, 07(01). https://doi.org/10.4172/2167-0234.1000314
- Nurokhmah, U., Sudarto, S., & Laksana, R. D. (2023). The effect of dividend payouts on firm value: The moderating role of institutional investors. *Sustainable Competitive Advantage (SCA)*, *12*(1), 1–9.
- Nympha, A. N., Uchenna, E., Priscilla, & Nnenna, C. V. (2021). Dividend Policy Determinants of Firm Value: Empirical Evidence from Listed Non-Financial Companies in Nigeria. *International Journal of Research and Innovation in Social Science*, *5*(7), 612–634.
- Obaidat, A. N. (2019). Is economic value added superior to earnings and cash flows in explaining market value added? An empirical study. *International Journal of Business, Accounting, & Finance, 13*(1).
- Obayagbona, J., & Ogbeide, D. (2018). Corporate taxes, agency costs and dividend policy of non-financial firms in Nigeria. *Amity Journal of Finance*, *3*(1), 61–86.
- Odum, A. N., Odum, C. G., Omeziri, R. I., & Egbunike, C. F. (2019). Impact of Dividend Payout Ratio on the Value of Firm: A Study of Companies Listed on the Nigerian Stock Exchange. *Indonesian Journal of Contemporary Management Research*, 1(1), Article 1. https://doi.org/10.33455/ijcmr.v1i1.84
- Ofori-Sasu, D., Abor, J. Y., & Osei, A. K. (2017). Dividend policy and shareholders' value: Evidence from listed companies in Ghana. *African Development Review*, 29(2), 293– 304.
- Olang, M. A., & Grace, A. M. (2017). Effect of working capital on the dividend pay-out by firms listed at the Nairobi securities exchange, Kenya. *International Journal of Finance and Banking Research*, *3*(2), 13–23.
- Olawale, L. S., Ilo, B. M., & Lawal, F. K. (2017). The effect of firm size on performance of firms in Nigeria. *Aestimatio: The IEB International Journal of Finance*, 15, 68–87.
- Onanjiri, R., & Korankye, T. (2014). Dividend payout and performance of quoted manufacturing firms in Ghana. *Research Journal of Finance and Accounting*, 5(15), 442–454.
- Oppong, F. K. (2016). *Dividend policy and firms' performance: A case of listed banks in Ghana* [Doctoral dissertation, Kwame Nkrumah University of Science and Technology]. http://ir.knust.edu.gh/bitstream/123456789/8651/1/OPPONG%20FOSU%20KENNE DY.pdf
- Ozuomba, C., Anichebe, A., & Okoye, P. (2016). The effect of dividend policies on wealth maximization–a study of some selected plcs. *Cogent Business & Management*, *3*(1), 1226457.

- Phornlaphatrachakorn, K., & Na-Kalasindhu, K. (2020). Strategic Management Accounting and Firm Performance: Evidence from Finance Businesses in Thailand. *The Journal of Asian Finance, Economics and Business*, 7(8), 309–321. https://doi.org/10.13106/JAFEB.2020.VOL7.NO8.309
- Pradana, I. Y. (2021). Effect of Leverage, Growth, Firm Size, Dividend Policy, and Interest Rate on Company Value. *Dinasti International Journal of Economics, Finance & Accounting*, 2(3), 316–327.
- Puspitasari, E., SUDIYATNO, B., HARTOTO, W. E., & WIDATI, L. W. (2021). Net interest margin and return on assets: A Case Study in Indonesia. *The Journal of Asian Finance, Economics and Business*, 8(4), 727–734.
- Putra, A. A. N. D. A., & Astika, B. (2019). The effect of corporate social responsibility disclosure on company value with liquidity as a moderation variable. *E-Journal of Accounting*, *26*(1), 30–57.
- Rafid, A. G., Pohan, H. T., & Noor, I. N. (2019). The effect of financial performance on firm value with disclosure of corporate social responsibility as a moderating variable. *Jurnal Akuntansi Trisakti*, 4(2), 245–258. https://doi.org/10.25105/jat.v4i2.4838
- Rahman, A. (2018). Effect of dividend policy on firm's performance: A case study of cement sector of Pakistan. SEISENSE Journal of Management, 1(5), 6–15.
- Rahman, M. M., Hamid, M. K., & Khan, M. A. M. (2015). Determinants of bank profitability: Empirical evidence from Bangladesh. *International Journal of Business and Management*, 10(8), 135.
- Rashid, C. A. (2018). Efficiency of financial ratios analysis for evaluating companies' liquidity. *International Journal of Social Sciences & Educational Studies*, *4*(4), 110.
- Restianti, T., & Agustina, L. (2018). The effect of financial ratios on financial distress conditions in sub industrial sector company. *Accounting Analysis Journal*, 7(1), 25–33.

- Rizqia, D. A., & Sumiati, S. A. (2013). Effect of managerial ownership, financial leverage, profitability, firm size, and investment opportunity on dividend policy and firm value. *Research Journal of Finance and Accounting*, 4(11), 120–130.
- Rizwan, M., Khan, M. N., Nadeem, B., & Abbas, Q. (2016). The impact of workforce diversity towards employee performance: Evidence from banking sector of Pakistan. *American Journal of Marketing Research*, 2(2), 53–60.
- Samo, A. H., & Murad, H. (2019). Impact of liquidity and financial leverage on firm's profitability – an empirical analysis of the textile industry of Pakistan. *Research Journal* of Textile and Apparel, 23(4), 291–305. https://doi.org/10.1108/RJTA-09-2018-0055
- Santosa, P. W. (2020). The moderating role of firm size on financial characteristics and islamic firm value at indonesian equity market. *Business: Theory and Practice*, *21*(1), 391–401.
- Sausan, F. R., Korawijayanti, L., & Ciptaningtias, A. F. (2020). The effect of return on asset

(ROA), debt to equity ratio (DER), earning per share (EPS), total asset turnover

(TATO) and exchange rate on stock return of property and real estate companies at Indonesia stock exchange period 2012-2017. *Ilomata International Journal of Tax and Accounting*, *I*(2), 103–114.

- Selvam, M., Gayathri, J., Vasanth, V., Lingaraja, K., & Marxiaoli, S. (2016). Determinants of firm performance: A subjective model. *Int'l J. Soc. Sci. Stud.*, 4, 90.
- Seth, R., & Mahenthiran, S. (2022). Impact of dividend payouts and corporate social responsibility on firm value–Evidence from India. *Journal of Business Research*, 146, 571–581.
- Setiadharma, S., & Machali, M. (2017). The effect of asset structure and firm size on firm value with capital structure as intervening variable. *Journal of Business & Financial Affairs*, 6(4), 1–5.

- Setiyawati, L., Wahyudi, S., & Mawardi, W. (2017). The influence of dividend policy, debt policy, independent commissioner, and institutional ownership on the firm value with growth opportunities as moderator variables (study on non-financial companies listed on IDX in the period of years of 2012-2015). *Jurnal Bisnis Strategi*, *26*(2), 146–162.
- Shah, S. A., & Noreen, U. (2016). Stock price volatility and role of dividend policy: Empirical evidence from Pakistan. *International Journal of Economics and Financial Issues*, 6(2), 461–472.
- Shao, L., Kwok, C. C., & Guedhami, O. (2010). National culture and dividend policy. Journal of International Business Studies, 41(8), 1391–1414.
- Singh, N. P., & Tandon, A. (2019). The Effect of Dividend Policy on Stock Price: Evidence from the Indian Market. *Asia-Pacific Journal of Management Research and Innovation*, 15(1–2), 7–15. https://doi.org/10.1177/2319510X19825729
- Sondakh, R. (2019). The effect of dividend policy, liquidity, profitability and firm size on firm value in financial service sector industries listed in Indonesia Stock Exchange 2015-
 - 2018
 period.
 Accountability,
 8(2),
 Article
 2.

 https://doi.org/10.32400/ja.24760.8.2.2019.91-101
- Suardana, I. B. R., Astawa, I. N. D., & Martini, L. K. B. (2018). Influential factors towards return on assets and profit change (study on all BPR in Bali province). *International Journal of Social Sciences and Humanities*, 2(1), 105–116.
- Sugiastuti, R., Dzulkirom, M., & Rahayu, S. (2018). Effect of profitability, leverage toward dividend policy and firm value. *Russian Journal of Agricultural and Socio-Economic Sciences*, 80(8), 88–96.
- Tahir, S. H., Sohail, S., Qayyam, I., & Mumtaz, K. (2016). Effect of corporate governance index on dividend policy: An investigation of textile industry of Pakistan. *Journal of Economic and Social Development*, 3(1), 139.

- Tanushev, C. (2016). Theoretical models of dividend policy. *Economic Alternatives*, *3*(1), 299–316.
- Triani, N., & Tarmidi, D. (2019). Firm value: Impact of investment decisions, funding decisions and dividend policies. International Journal of Academic Research in Accounting, Finance and Management Sciences, 9(2), 158–163.
- Ugwu, C. C., Onyeka, V. N., & Okwa, I. E. (2020). *Dividend Policy and Corporate Financial Performance: Evidence from Selected Listed Consumer Goods Firms in Nigeria* (SSRN Scholarly Paper 3672279). https://papers.ssrn.com/abstract=3672279
- Utami, W. B. (2017). Analysis of current ratio changes effect, asset ratio debt, total asset turnover, return on asset, and price earning ratio in predictinggrowth income by considering corporate size in the company joined in IQ45 index year 2013 -2016.
 International Journal of Economics, Business and Accounting Research (IJEBAR), 1(01), Article 01. https://doi.org/10.29040/ijebar.v1i01.253
- Wells, P. (2016). Economies of scale versus small is beautiful: A business model approach based on architecture, principles and components in the beer industry. Organization & Environment, 29(1), 36–52.
- Widyastuti, M. (2019). Analysis of liquidity, activity, leverage, financial performance and company value in food and beverage companies listed on the Indonesia Stock
 Exchange. SSRG International Journal of Economics and Management Studies (SSRG-
- *IJEMS*), 6(5), 52–58.
 Yuniningsih, Y., Pertiwi, T., & Purwanto, E. (2019). Fundamental factor of financial management in determining company values. *Management Science Letters*, 9(2), 205–216.
- Zuhroh, I. (2019). The effects of liquidity, firm size, and profitability on the firm value with mediating leverage. *"Sustainability and Socio Economic Growth"*, 2019, 203–230.

https://doi.org/10.18502/kss.v3i13.4206

APPENDICES						
APPENDIX I: List of Selected Firms						
Ecobank Ghana Limited						
Anglogold Ashanti						
Access Bank						
Benso Oil Palm Production						
Cal Bank						
Enterprise Group Limited						
Ecobank Transactional Incorporated						
GCB						
Guiness Ghana Brewery Limited						
MTN Ghana						
Republic Bank Ghana PLC						
Goil Ghana PLC						
Total Energies						
Standard Chartered Bank Limited APPENDIX II: Similarity Index						

Impact of dividend policy on financial performance and firm value

ORIGINALITY REPORT						
2 SIMILA	0%	16% INTERNET SOURCES	12% PUBLICATIONS	9% STUDENT PAPERS		
PRIMARY	SOURCES				_	
1	WWW.res Internet Source	earchgate.net		2	.%	
2	ir.jkuat.ad	c.ke		1	%	
3	phd-dissertations.unizik.edu.ng				%	
4	Perdana Wahyu Santosa, Ovinda Aprilia, Martua Eliakim Tambunan. "The Intervening Effect of the Dividend Policy on Financial Performance and Firm Value in Large Indonesian Firms", International Journal of Financial Research, 2020 Publication				%	
5	5 krishikosh.egranth.ac.in Internet Source				%	
6	"From th of Ideas: Springer 2023 Publication	e Internet of Th The Role of Ar Science and Bu	hings to the In tificial Intellige usiness Media	ternet 1 ence", LLC,	%	